

Complements and Substitutes in Sequential Auctions: The Case of Water Auctions

Donna, Javier and Espin-Sanchez, Jose

Ohio State University, Northwestern

22 July 2015

Online at https://mpra.ub.uni-muenchen.de/66997/ MPRA Paper No. 66997, posted 02 Oct 2015 10:15 UTC

COMPLEMENTS AND SUBSTITUTES IN SEQUENTIAL AUCTIONS: THE CASE OF WATER AUCTIONS^{*}

and

Javier Donna

José-Antonio Espín-Sánchez

The Ohio State University

Yale University

First version: December 10, 2010 This version: July 22, 2015

^{*}We are indebted to Joel Mokyr and Rob Porter for helpful discussions, guidance, and support. Discussions with Meghan Busse, Regina Graffe, Aviv Nevo, Florian Zettelmeyer, Jason Blevins, Jia-Young (Mike) Fu, Matt Gunden, Jim Peck, Bill Rogerson, Ron Siegel, Alex Torgovitsky, Greg Veramendi, and comments from an anonymous referee, as well as seminar participants at the International Industrial Organization Conference (2013), the Jornadas de Economia Industrial (XXVIIth edition), the Meetings of the European Association for Research in Industrial Economics (EARIE 2012), the North American Summer Meetings of the Econometric Society (2012), the Second Annual UTDT Economics Conference, the World Economic History Congress (XVIth edition), the Midwest Economics Association Annual Meeting (2013), Aarhus University, Analysis Group, Arizona State University, Bates White, Charles River Associates, Centro de Estudios Monetarios y Financieros (CEMFI), Compass Lexecon, Johns Hopkins University, Northwestern University (Industrial Organization and History Workshops), The Ohio State University, and Universidad Carlos III have greatly benefited this work. We would also like to express our gratitude to Fernanda Donna and Antonio Espín for superb research assistance, to the librarians from the archive of Mula for valuable help in accessing the historical files, and to Maja Butovich, Kelly Goodman, Elizabeth Lenaghan, and Melissa Petruzzello for editorial advice. We thank the AEMET for providing us with the meteorological data. José-Antonio Espín-Sánchez also acknowledges financial support from Fundación Caja Madrid. A version of this paper was part of Javier Donna's Ph.D. dissertation at Northwestern University. All errors are involuntarily. The online appendix for this paper is http://www.jdonna.org/water-auctions-web.

Abstract

We study sequential auctions in which bidders demand multiple units. We collect a novel data set on sequential water auctions for the empirical study. Although water units are identical, two features from the empirical setting create a trade-off whereby units of water end up being complements or substitutes. First, there is a water loss that is only incurred for the first unit, generating a sunk cost. Second, subsequent units of water exhibit decreasing marginal returns. Units of water are complements or substitutes depending on the relative importance of the sunk cost and decreasing returns. Weather seasonality provides us with the required variation (in sunk costs relative to decreasing returns) to perform the empirical investigation. When units are complements, one bidder wins all units by paying a high price for the first unit, thus deterring others from bidding on subsequent units. When units are substitutes, different bidders win the units with positive probability and pay prices of similar magnitude, even when the same bidder wins all units. We analyze this stark pattern of outcomes not investigated in the literature before. We recover individual demand consistent with this pricing behavior and confirm it is not collusive, but consistent with non-cooperative behavior. Demand estimates are biased if one ignores these features.

JEL CODES: D44, C13, L10, L40

KEYWORDS: Auctions, Structural Demand Estimation, Market Structure, Competition, Collusion

Javier Donna Department of Economics The Ohio State University 1945 N High St, 417 Arps Hall Columbus, OH 43210 Phone: 614-688-0364 Email: donna.1@osu.edu José-Antonio Espín-Sánchez Department of Economics Yale University 27 Hillhouse Ave New Haven, CT 06511 Phone: 203-432-0890 Email: jose-antonio.espin-sanchez@yale.edu

1 Introduction.

There are many instances in the real world where several units of the same or similar goods are allocated sequentially or periodically using auctions. Examples include timber, procurement of public goods, electromagnetic spectrum, and treasury bills. The nature of the goods at auction and the firms bidding determine whether the goods are complements (increasing marginal returns) or substitutes (decreasing marginal returns). In many cases, goods are complements because firms incur fixed costs to realize the full value of purchased goods. This is the case of the machinery and workers needed to fell trees or build highways. Firms also experience decreasing marginal returns due to limited capacity to hire more workers or buy more machinery. Decreasing marginal returns also arise as a consequence of the downward sloping demand for the firms' final products. Once a firm has a valid spectrum for a given county, the value of another tranche of the spectrum decreases substantially. We expect firms to have increasing marginal returns if the first effect dominates, decreasing marginal returns if the second effect dominates, and hill shaped marginal returns if both effects are important.

By affecting the valuation of subsequent units, fixed costs and decreasing returns determine bidder behavior and price dynamics. Price dynamics are central to connect observed bids to the underlying distributions that characterize individual demand, which is fundamental to discuss positive and normative questions. For instance, variation in prices caused by a high sunk cost will affect even relatively simple tasks such as measuring the dispersion in individuals' private valuations. Moreover, in such a case, a competitive environment could be incorrectly interpreted as collusive.

The existing literature on sequential auctions has provided little empirical evidence on the effect that complementarities or substitutabilities in the valuation of subsequent units has on price behavior. The main reason for this lack of evidence is the challenge of finding sufficient variation in the degree of complementarity. Our aim is to address this empirical gap. To that end, we examine a unique panel data set that exploits large changes in the degree of complementarity across seasons: variation in the importance of sunk costs relative to decreasing returns. We use this variation to analyze bidding behavior in sequential auctions in which buyers' preferences for multiple units exhibit both sunk costs and decreasing marginal returns. We investigate its implications for price dynamics and price competition.

The data in this paper comes from sequential water auctions from a self-governed community of farmers in Mula, Spain. The data allows to exploit a unique scenario to analyze a stark pattern of outcomes not previously documented in the literature. Sometimes, winning prices exhibit a standard competitive pattern. In this scenario, winning prices are similar in magnitude, regardless of whether the same or different bidders (farmers in our case) win the sequential units.¹ Other times, one farmer wins all the units, pays a high price for the first unit, deters other farmers from entering subsequent auctions, and thus pays a very low price

¹Declining prices for identical objects is an empirical regularity known as the *declining price anomaly*, which was first documented by Ashenfelter (1989) in his seminal paper.

for the remaining units. We call this the *deterrence* effect. We show that this pattern of outcomes is consistent with a non-cooperative equilibrium, where the observed price dynamics are competitive, not collusive.

The data for our analysis consist of individual winning bids and auction covariates. These covariates include the amount of rainfall. The basic unit of sale is the right to use three hours of water (432,000 liters) for irrigation. For each weekday, eight units are sold for each schedule: four for daytime (7AM-7PM) and four for nighttime (7PM-7AM) irrigation. The auctioneer sells first the twenty units corresponding to nighttime and then the twenty units corresponding to daytime. This leaves ten four-unit sets of auctions that are sold in order. (The ten sets of four-unit auctions are: Monday-nighttime, Tuesday-nighttime, and so on until Friday-daytime.) Thus, the relevant unit of analysis for investigating individuals' demand and the pattern of outcomes is four-unit auctions. But units within each four-unit set are not conditional-independent due to the presence of sunk costs. Observing the winner's identity allows us to estimate the model, as outlined in Section 6. Local weather conditionally, as less rain falls in summer than in winter in southern Spain, the presence of seasonalities provides us with the variation in sunk costs relative to decreasing returns necessary to perform the empirical investigation.

The interpretation of the data based on our economic model is fundamental to our approach. We model the environment as a sequential (ascending price) English auction along the lines of Engelbrecht-Wiggans (1993) and von der Fehr (1994) in which bidders, by incurring a participation cost, decide whether to attend each sale. We focus on the symmetric conditional-independent private values paradigm that has dominated the literature (Donald and Paarsch 1996). We incorporate two features from our empirical setting. First, a sunk cost is incurred for the first unit bought because water flows through a channel dug into the ground. Some water is lost when the channel is dry (the first unit), but the loss is negligible for subsequent units. Engineers have estimated that 20% of the water of the first unit that travels through a dry channel was lost (González-Castaño and Llamas-Ruiz 1991). Second, decreasing marginal returns are present for subsequent units because the amount of irrigated land is fixed.

The relative importance of sunk costs and decreasing marginal returns generates a tradeoff, whereby buyers' bidding behavior depend on whether different units are complements or substitutes. When goods are complements, the same bidder wins all the objects paying a high price for the first unit equal to the valuation for the whole bundle (*four* times the second highest valuation for the first unit, adjusted for the complementarity effect and participation cost). By doing this, the winner of the first unit deters others from bidding on the remaining three units, allowing this bidder to pay very low prices (close to zero) for the remaining three units. The resulting price pattern, along with the same bidder winning all the units, may lead to an incorrect collusive interpretation. When goods are substitutes, different bidders win the objects with a positive probability and pay prices of similar magnitude, even when the same bidder wins all the objects. We provide empirical evidence for the key features of our model: participation and sunk costs. We argue that bidders are better informed than the seller, whose mechanism ignores bidder preferences for multiple units. Nevertheless, a sequential English auction achieves ex-ante efficiency, as we discuss in Subsection 7.2.

The price patterns that our model predicts provide us with a straightforward empirical method to determine the regime being played (complements or substitutes). When goods are complements, very low prices are paid by the same winner (the winner of the first unit) for the second, third, and fourth units. This allows us to separate the data for each regime using the procedure outlined in Section 5.

We estimate the distribution of private valuations by maximum likelihood using an exponential distribution and the English structure for the auction. To estimate sunk cost and decreasing marginal returns, we form moment conditions based on the structural equations of the model. We infer participation costs using data from auctions in which bidders were present, but no one placed bids. This method gives us bounds on participation costs.

Our empirical work establishes three main results. First, we recover individual demand—characterized by private valuations and the model's structural parameters—that is consistent with the described price patterns and the *deterrence* effect in particular. Second, the equilibrium price dynamics are consistent with competitive behavior. Non-cooperative behavior is not only consistent with the *deterrence* effect, but also predicts such price differentials. Incentives to deviate from a collusive strategy are higher in spring and summer, when water is more valuable. However, it is in spring and summer when we observe noncooperative behavior more often. Finally, we show that estimates that ignore the importance of participation and sunk costs will be biased. We test whether price variations, conditional on covariates, are better explained by our proposed model or a standard English auction model without participation costs, using that the latter is encompassed by the former. The approach of Haile and Tamer (2003), that relies on two basic behavioral assumptions, provides a robust structural framework for inference. These minimal assumptions are not satisfied in the present context. We discuss how Haile and Tamer's structure can be interpreted in the current setting.

In the next section, we describe the related literature. Section 3 discusses the auction system, the empirical regularities, and the modeling assumptions required in our context. Section 4 presents the model. Section 5 discusses regime determination. Section 6 examines the estimation procedure. Section 7 presents the results, analyzes the importance of sunk costs, and the interpretation of complementarities. Finally, Section 8 concludes. Additional data description is provided in Section A in the online appendix. All proofs and extensions of the model are in Section B in the online appendix.

2 Contributions and Related Literature.

In this section we describe the related literature and highlight how our paper contributes to the current body of work. This paper is most similar to the empirical literature investigating the predictions of strategic bidding in sequential auctions with multi-unit demand. To best of our knowledge, the price dynamics that we investigate (see Section 3) have not been documented in the literature before. Most of the literature do not consider participation costs in their analysis. We show that participation costs affect equilibrium outcomes. We then use our model to partially identify participation costs and estimate informative bounds.

Numerous empirical studies have highlighted the importance of complementarities (Anton and Yao 1987; Gandal 1997; Wolfram 1998; Pesendorfer 2000; Marshall, Raiff, Richard, and Schulenberg 2006).² Substitutabilities are a major component in several industries such as sequential highway construction procurement auctions (Jofre-Bonet and Pesendorfer 2003), sequential timber auctions (List, Millimet, and Price 2004), or sequential cattle auctions (Zulehner 2009). Several authors have studied cases of either complements due to synergies among auctioned goods, or substitutes due to decreasing marginal utility (Black and De Meza 1992; Branco 1997; Liu 2011). Selling goods in a bundle increases a seller's revenue when goods are complements (Palfrey 1983; Levin 1997, Armstrong 2000). Our setting differs from these scenarios in that we consider sequential, instead of simultaneous, auctions.³

Prior investigations of the relationship between sequential auctions and the complementarity or substitutability between identical units are more scarce (*e.g.* Jeitschko and Wolfstetter 2002; Jofre-Bonet and Pesendorfer 2012). Jeitschko and Wolfstetter 2002 analyze optimal sequential auctions in a binary-valuations case. They find that English auctions extract more rent than first-price auctions. Our model differs as we consider the class of continuous valuation distributions. Jofre-Bonet and Pesendorfer 2012 allow for complementarities and substitutabilities in a model of sequential auctions. They find that while first-price auctions give greater revenue than second-price (English) auctions when the goods are substitutes, the opposite is true for complements. Both mechanisms are efficient in their model. Their predictions about price trends are consistent with previous findings. Contrary to our analysis with participation costs, where buyers are better informed than the seller, Jofre-Bonet and Pesendorfer 2012 examine the effect of capacity constraints on bidding behavior in procurement auctions using a two-period auction game where sellers have private information about their costs. Balat (2013) and Groeger (2014) analyze dynamic auctions in the highway procurement market. Balat (2013) extends the model from Jofre-Bonet and Pesendorfer 2012 by

²Outside the auction literature, Gentzkow (2007) studies the value of new goods using a model encompassing the possibility of both complementarities and substitutabilities.

³See Milgrom (2000) and Ausubel (2004) for recent contributions to this literature. Edelman, Ostrovsky, and Schwarz (2007) study the properties of a "generalized English auction" used to sell Internet advertisements and show their proposed mechanism has a unique equilibrium. Kagel and Levin (2005) experimentally investigate multi-unit demand auctions with synergies, and compare behavior in sealed-bid and ascending-bid uniform-price auctions. See Kagel (1995) for a survey on laboratory experimental auction markets.

allowing endogenous participation and unobserved heterogeneity. Groeger (2014) analyzes bidder learning in the entry stage of an auction game.⁴ Finally, Hendricks and Porter (1988) conducted an early and influential investigation on how interdependencies among auctioned objects affect the auction's outcome. They analyze auctions for drainage leases and show that better informed firms (which hold tracts neighboring the drainage tracts that were auctioned) earned higher rents than uninformed ones.

This paper makes a methodological contribution by developing an empirical model of sequential English auctions with participation costs that allows units to complement and substitute for the *same* bidder depending on seasonalities. The model produces distinguishable price pattern predictions in each regime. This feature allows us to determine the regime under which the game is being played using end-digit preferences. This allows us to weaken the behavioral assumptions, such as the specification of bidders' beliefs, that would be necessary to solve the whole game (see Sections 4 and 5). Similar to the work of Hendricks and Porter (1988) and Haile (2001), we show evidence inconsistent with the equilibrium predictions of standard models and supportive of a model that captures sunk costs, decreasing marginal returns, and participation costs. Not accounting for these features may lead to the incorrect interpretation of a competitive market as collusive.

We build upon the existing literature on participation costs and entry fees (McAfee and McMillan 1987; Engelbrecht-Wiggans 1993; von der Fehr 1994) by constructing a sequential English auction model similar to that of Von der Fehr. However, our set-up differs in that bidders are allowed to buy more than one unit of the good. von der Fehr 1994 considers the case when goods are independent and finds the same equilibrium as that of our complementarities case.

While the auction literature has studied price dynamics and the relationship between sequentially auctioned goods (for example, Weber 1983; McAfee and Vincent 1993; Benhardt and Scoones 1994; Engelbrecht-Wiggans 1994), to the best of our knowledge, we analyze a stark pattern of outcomes not investigated in the literature before. Sometimes, when goods are substitutes, winning prices exhibit a standard competitive pattern: regardless of whether the same or different bidders win the sequential units, winning prices are similar in magnitude. Other times, when goods are complements, the same bidder wins all units by paying a high price for the first unit, deterring others from bidding on subsequent units.⁵ We show that this pattern of outcomes is consistent with a competitive market structure. We are not aware of any study where identical units may complement *and* substitute within the same market and for the *same* bidder.⁶ In addition to recovering the structural parameters that characterize

⁴In contrast, we investigate a stark price dynamics not documented before (see Section 3). In our empirical setting, the same identical units sometimes complement and other times substitute for the same bidder. We show that these price dynamics are not collusive, but consistent with non-cooperative behavior. In addition, we infer participation based on two simple assumptions that provide us informative bounds.

⁵Declining or downward price trends in sequential auctions, the results we describe in Subsection 3.3, have been broadly documented (for example, Ashenfelter 1989; Ashenfelter and Genesove 1992; McAfee and Vincent 1993).

⁶The literature in multi-unit auctions can be divided into sequential auctions, in which the auctioneer

individual demand and confirming it is consistent with non-cooperative behavior, which are of interest to the literature on empirical auctions, we collect a unique panel data set to examine a market institution that was active and stable for eight centuries in a self-governed community of farmers in southern Spain.⁷ Understanding this strategic non-cooperative behavior of bidders in this stable market institution is of independent interest.

3 Background on the Market.

The data in this paper come from all water auctions in Mula, Spain, from January 1954 through August 1966, when the last auction was run.⁸ On August 1st, 1966, the allocation system was modified from an auction to a two-sided bargaining system. In the bargaining system, the *Heredamiento de Aguas* (water-owners holding) and *Sindicato de Regantes* (land-owners association) arranged a fixed price for every *cuarta* of water (the smallest unit auctioned). Gradually, the *Sindicato de Regantes* bought shares in the *Heredamiento de Aguas* association until they finally merged in 1974. Thereafter, water was allocated to each farmer following a fixed quota with each piece of land entitled to some proportion of the water every year.

The reasons for focusing on the period from 1954 to 1966 are, first, that it represents the final period of the auction allocating system in use for at least eight centuries in this region. Second, the government conducted a special agricultural census in 1954/55, providing detailed information about the farmers who bid in this period's auctions

The study of these sequential auctions introduces a unique circumstance for analyzing a stark pattern of outcomes not previously documented in the literature. Sometimes, winning prices exhibit a standard competitive pattern where, regardless of whether the same or different farmers win the sequential units, prices are similar in magnitude (Figure 1). Other times, one farmer wins all sequential units: he pays a high price for the first unit, *deterring* other farmers from entering subsequent auctions, thus paying a very low price for the remaining units (Figure 2).⁹ This stark pattern of outcomes is consistent across the whole sample (see

sells the units following a series of sequential steps using a single-unit auction each time, and simultaneous auctions, in which the auctioneer uses a complex mechanism to allocate all units simultaneously. For recent contributions see Kastl (2011), who investigates bidders submitting step functions as their bids in multi-unit treasury bills auctions, and Reguant (2013), who studies complementarity bidding mechanisms used in wholesale electricity auctions. Implementing a simultaneous auction requires a strong commitment from the auctioneer either to not renege in the promised mechanism, or to use the information elicited in the process to demand a higher price for the good. This also imposes technical difficulties in the way bidders frame their contingent bids (Cramton, Shoham, and Steinberg 2006). Neither of these conditions are satisfied in our setting. Hortaçsu (2011) discusses recent progress in the empirical study of multi-units auctions. See Kagel and Levin (2001) for an experimental investigation when bidders demand multiple units in sealed bid and ascending auctions.

⁷In the lead article of the first issue of the *American Economic Review*, Coman (1911) provides an early discussion of the same institution that is analyzed in detail in this paper. For an extensive study of self-governed irrigation communities see Ostrom (1992).

⁸Data available in the historical archive of Mula go back to 1803.

⁹In terms of purchasing power, one peseta from 1950 is approximately equivalent to 0.43 U.S. dollars from

Table 1 and Figure 5 that we describe in Subsection 3.3).

3.1 Water Auctions as Allocation System.

Although the process of allocating water in Mula has varied slightly over time, the basic structure has essentially remained unchanged since the 15^{th} century. Land in Mula is divided into *regadío* (irrigated land) and *secano* (dry land). Irrigation is only permitted in the former. A channel system directs water from the river to *regadío* lands.¹⁰ *Regadío* are fertile lands close to rivers, and thus allow a more efficient use of the water in the region. Since it is forbidden to irrigate lands categorized as *secano*, only the farmers that own a piece of *regadío* land in Mula are allowed to buy water.

The mechanism to allocate water to those farmers was a sequential outcry ascending price (or English) auction. The auctioneer sold by auction each of the units sequentially and independently of each other. The auctioneer tracked the name of the buyer of every unit and the price paid by the winner.¹¹

The basic selling unit is a *cuarta* (quarter), which is the right to use water that flows through the main channel for three hours. Water storage is done in the *De La Cierva* dam. Water flows from the dam through the channels at approximately 40 liters per second. As a result, one *cuarta* carries approximately 432,000 liters of water. Traditionally, auctions were held every 21 days to complete a *tanda* (quota), the basic aggregate unit of irrigation time. During our sample period, auctions were carried out every Friday.

During each session, 40 *cuartas* were auctioned: four *cuartas* for irrigation during the day (from 7:00 AM to 7:00 PM) and four *cuartas* for irrigation during the night (from 7:00 PM to 7:00 AM), for each weekday (Monday to Friday). The auctioneer first sold the 20 *cuartas* corresponding to the night-time, and then the 20 *cuartas* corresponding to the day-time. Within each day and night group, units were sold beginning with Monday's four *cuartas*, and finishing with Friday's.

3.2 The Dataset.

We combine data from four sources. The first is auction data, that we collected from the historical archive of Mula.¹² Based on bidding behavior and water availability, auction data

^{2013 (}for details see Section A in the online appendix).

¹⁰The channel system was expanded from the 13^{th} to 15^{th} century as a response to the greater demand for land due to population increase. The *regadio* land structure has not changed since the 15^{th} century.

¹¹The farmers could not store water in their plots. Reselling water was forbidden. While a farmer could steal water by opening the gate next to his own parcel, the technology for detection of this crime was effective as irrigation was done by flood irrigation (more on this in Subsection 4.2). It was easy to determine who stole water just by identifying a flooded parcel from a farmer who did not buy water in the auction for that specific day-schedule (conditional on rainfall). The *Tribunal de los Hombres Buenos* (Council of Good Men), composed by elected members among the farmer community, was responsible to adjudicate conflicts between the farmers. Conflicts mostly arose over unpermitted irrigation. We investigate this behavior in Donna and Espin-Sanchez (2013b).

¹²From the section *Heredamiento de Aguas*, boxes No.: HA 167, HA 168, HA 169, and HA 170.

can be divided into three categories: (i) Regular periods, when the name of the winner, price paid, date and time of the irrigation for each auction transaction was registered; (ii) No-supply periods, when no auctions were conducted due to water shortage in the river or damage to the dam or channels, usually due to intense rain; and finally (iii) No-demand periods, when auctions were held but no one bid, leaving the registration auction sheet blank. The sample for this study includes nearly 13 years of auction data spanning January 1954 to August 1966. Every week, 40 units (corresponding to 40 *cuartas*) were sold, with the exceptions being when no auction was run (no-supply) or no bids were observed (no-demand). A total of 17,195 auctions were run during the period under analysis.¹³

We link auction data to the data that we collected from the 1954/55 agricultural census from Spain, which provides information on individual characteristics of farmers' land.¹⁴ The census was conducted by the Spanish government to enumerate all cultivated soil, production crops, and agricultural assets available in the country. Individual characteristics for the farmers' land (potential bidders which we link with the names in the auctions data) include the type of land and location, area, number of trees, production, and the price at which this production was sold in the census year. Figure 3 shows a sample card for one farmer from the census data. During the 13-year period under analysis, there were approximately 500 different bidders in our sample. The number of bidders who won auctions during a specific year was considerably lower—the mean for our sample is around 8 (see table 5, discussed in Subsection 6.2)—and conditional on participation, each farmer won on average 22 units per year. This is consistent with the census data, where mean land extension is 5.5 ha. with an average of 33 trees per ha.¹⁵

We also link auction data to daily rainfall data for Mula and monthly price indices for Spain, which we obtain from the *Agencia Estatal de Metereología*, AEMET (the National Meteorological Agency), and the *Instituto Nacional de Estadística de España*, INE (the National Statistics Institute of Spain), respectively.

¹³Table A1 on page A-3 in the online appendix displays the frequency distribution of units in the auctions disaggregated by the units bought sequentially by the same farmer.

¹⁴From the section *Heredamiento de Aguas* in the historical archive of Mula, box No. 1,210.

¹⁵Average annual rainfall during the period is 320 mm. Recent irrigation studies on young citrus plantings have shown a water use of 2-5 megalitres per hectare annually (Chott and Bradley 1997). Water savings are possible if irrigation can be allocated to similar units of production, such as young trees or reworked sections of a property. In arid regions, like Murcia, water requirements are around 20% less and they are lower for mature trees. Some farmers that are part of water-owner holding use their own water instead of selling it through auctions. Although water stress during droughts affects the quality of production, trees would hardly die as a result. During a normal year without drought, trees could survive the whole year from rainfall alone. For further details see, for example, Chott and Bradley (1997), Wright (2000), and du Preez (2001). Finally, note that although the average number of trees per farmer is 161 (see Table A2 in Subsection A in the online appendix), the average number of trees per hectare in our sample is 33, a lower number compared to the conventional agricultural standard spacing for citrus trees that is 100 trees per hectare.

3.3 Summary Statistics.

Mediterranean climate rainfall occurs mainly in spring and autumn. Peak water requirements for the products cultivated in the region are reached in spring and summer, between April and August. Soaring demand is reflected by the frequency of *auctions where the same farmer buys all four consecutive units* (4CU), which reaches its peak during these months (see Figure 14 and the discussion in Subsection 7.2). The frequency of 4CU is not homogenous over time, but is related to seasonal rainfall, as can be seen in Figure 4. Overall, 42% of the units were sold in 4CU.¹⁶ There are no observations where the same farmer buys more than four consecutive units, nor observations where the same farmer buys consecutive units across days (*e.g.* there are no observations where the same farmer buys the last units of a day-auction, and the first units of the night-auction).

We only observe the transaction price (winning bid) and the identity of the winner (name). (We do not observe all bids.) There is substantial price variation, both within and across four-unit auctions. Winning prices range from 0.05 pesetas (*ptas*) to 4,830 *ptas*, with a mean of 271.6 *ptas*. As expected, winning prices and the frequency distribution of 4CU are strongly correlated with past rainfall (Figure 4). Table 1 exhibits the distribution of winning prices by both the number of consecutive units bought by the same individual (1CU, 2CU, 3CU, or 4CU) and by sequential auction $(1^{st}, 2^{nd}, 3^{rd}, \text{ or } 4^{th})$. The greater variation that we observe for 4CU (with respect to non-4CU) has a well defined pattern. While mean prices for the first auction in 4CU are considerably higher than for non-4CU (Table 1, Panel 2: 677.6 *ptas* for 4CU against 211.1 *ptas* for 1CU, 305 *ptas* for 2CU, or 410 *ptas* for 3CU), mean prices for fourth auctions in 4CU are the smallest (Table 1, Panel 5: 210.1 *ptas* for 4CU against 233.4 *ptas* for 1CU, 239.6 *ptas* for 2CU, or 311.6 *ptas* for 3CU). Median and maximum prices display similar patterns.

Figure 5 presents price variation by number of consecutive auctions won by the same individual (left panel) and by sequential auction (right panel). The figure shows that the stark pattern of outcomes from Figures 1 and 2 is consistent across the whole sample. On the one hand, in the top panel of Figure 5 we can see that price dispersion—as well as the mean and median price—is higher when the same farmer wins all four consecutive units (in the top panel, last vertical box labeled 4). On the other hand, in the bottom panel, where we further disaggregate each box from the top panel by unit (first unit, second unit, third unit, and fourth unit), we can see that the higher price dispersion for unit 4 in the top panel—as well as the higher mean and median—is generated by the greater variation in prices for first units in the lower panel (not by prices in second, third, and fourth units).

This particular pattern in prices is caused by the above mentioned *deterrence* effect whereby farmers exhibit different behavior based on seasonality and rainfall, *i.e.*, residual demand for water. During high demand and low rainfall months, the same farmer buys

¹⁶Table A1 in Section A in the online appendix displays the frequency distribution of units sold by number of units bought by the same farmer.

all four sequential units, paying a high price for the first unit (with respect to the median or average price conditional on rain) and very low prices for the remaining units. During months when demand is not high due to farming seasonalities or when rainfall is high, winning prices for all units are similar in magnitude, regardless of whether the same farmer wins all sequential units (4CU) or different farmers win subsequent units (1CU, 2CU or 3CU).

Aggregate prices over time display consistent trends with the ones found in the empirical literature on sequential auctions. Figure 6 shows that, on average, per unit prices decline by sequential unit (being the first unit of each day higher than second to fourth units), and by day of the week (prices decline from Monday to Friday). Figure 6 also shows that per unit prices are slightly higher during the day than during the night. High water requirements for citrus during summer causes prices to soar during those months (Figure 7). As expected, prices are also higher during droughts, after conditioning on seasons (Figure 8).¹⁷

Table 2 shows that these correlations are robust after conditioning on past rain, unit, weekday, schedule, week-of-the-year, month, and individual fixed effects. The table displays the results obtained by regressing daily unit prices on a seven-day-rain moving average (Rain $MA\gamma$), the rain on auction day, and the mentioned fixed effects. The estimated coefficients on Rain $MA\gamma$ have the expected sign and are statistically significant at the 1% level. From column 1, a 10 millimeter (mm) increase in average rain in the previous week is associated with a decrease of 40.5 ptas in the equilibrium price paid in the auction. The regression in column 2 adds unit, weekday, and schedule fixed effects. The estimated coefficient on Rain $MA\gamma$ increases in magnitude and also has the expected sign. This regression also shows that, as noticed in previous figures, price declines within day and across units (both for day-time and night-time auctions) and across schedules (price is on average 110 ptas lower for nighttime auctions than for day-time auctions). The estimated coefficients show that equilibrium prices decline monotonically within the week (Figure 6). Columns 3 and 4 add, respectively, month seasonal dummies and individual fixed effects (we have 537 different individuals in our sample) to the specification in column 2. The estimated coefficient on Rain MA7 in column 3, though smaller, again has the expected sign and is statistically different from zero. Similar qualitative results are obtained in column 4; however, the estimated coefficient on Rain $MA\gamma$ has increased. Note that the goodness of the fit in the last regression is 36%. indicating that average (or *ex-ante*) prices are explained relatively well by observables such as rain in the previous week and time of the allocation. This evidence supports the idea the observable (common knowledge) components of prices in drives four-unit auctions. Although not reported, we performed an analogue analysis using average daily prices within schedule as a robustness exercise and obtained similar results.

¹⁷See the online appendix for a discussion on droughts.

4 The Model.

As noted above, bidding behavior is a result of a complex decision process. There are three main features from the empirical setting that need to be accounted by the model: (i) sunk costs that farmers incur when they buy their first unit, (ii) decreasing marginal returns of subsequent units of water, and (iii) participation costs of farmers in this market.

Sunk Costs (SC). Water is allocated during the auction and is distributed on the specific day and time of the irrigation accordingly. Water stored in the dam is delivered to the farmer's plot on this date using the channel system. Except the main canal, all channels are dug into the ground (Figure 9). On the day of the irrigation, a guard opens the corresponding gates to allow the water to flow to the appropriate farmer's land. These channels are land-specific in the sense that different areas and lands have their own system of channels which only carry water when the corresponding gates are opened. A concern is that farmers whose lands lie next to each other may be buying different sequential units for the same auction. In this case, the SC would only be incurred by the first farmer for his first unit but not for the second farmer for his first unit. We use data on the specific location of the farmers that we match to auction winners to analyze these situations in Subsection 7.2. There is a water loss that is incurred because water flows over a dry channel. Engineers have estimated this loss to be between 15% and 40% (20% on average) of the water carried by one *cuarta* when the channel is completely dry (see Vera Nicolás 2004). This is the SC incurred by the bidder for his first unit. The SC is only incurred once, for the first unit, since water losses associated with a wet channel are negligible. The channel dries out after approximately 12 hours without water (González-Castaño and Llamas-Ruiz 1991). In 1974 the system of sub-canals was made of concrete, instead of just dug in the ground, to prevent such losses (González-Castaño and Llamas-Ruiz 1991).

Decreasing Marginal Returns (DMR). The second feature refers to the decreasing marginal returns (DMR) effect. The classic textbook case for DMR is appropriate for our empirical application. Given that the amount of land owned by each farmer is fixed, marginal productivity of subsequent units of water is decreasing. When assessing the relative importance of DMR, the impact in summer would generally be greater than in autumn. More generally, one would expect DMR to be affected by season and rain. When water requirements are high, the slope of the marginal productivity function will be relatively flat, as in the left panel in Figure 10. This is likely to occur in spring and summer. On the other hand, when water requirements are low, the slope of the marginal productivity function will be steeper, as in the right panel in Figure 10. This is likely to happen in autumn or winter.

Participation Costs. There are several reasons why farmers face participation costs in this market. The first component is opportunity cost. Farmers who value their time may

prefer not to participate in the whole auction session. Auctions were run on Fridays during work hours. Attending the auction entailed alternative use of working time for the farmer. This type of cost affects the number of farmers that participate in the auction, which we do not observe, but not the behavior of the farmer during the auction.

The second component of participation costs correspond to the hassle costs associated with active bidding. Only a fraction of the individuals who attended a Friday auction were actively engaged in the bidding for a particular sequential auction of water and not everyone who was present participated in every auction (Botía, Francisco, personal interview, Murcia, June 17, 2013).¹⁸ As von der Fehr (1994) points out, a reasonable assumption for why only a portion of attendees participate may be that they consider it so unlikely to that they will win at a price below what they will be willing to pay, that they are not willing to bother to engage in bidding. We expect this type of costs to be very small but positive.¹⁹

Empirical evidence from our data is consistent with the assertion that farmers dislike participation, facing positive entry costs as they do. We observe multiple weeks per year when auctions were run, farmers showed up and bought the first units of water, but no one bid for the last units. Since there was no reservation price and the minimum bid increment was cents, they could have potentially won all the remaining units bidding one cent. To the contrary, they decided not to bid and instead left the auction. For example, on January 22, 1954, units 1 to 16 were sold to seven different farmers but no one bid for units 17 to 20 (Figure 12). In 1954 we observe similar behavior for 14 weeks,²⁰ and this is consistent along the remaining years in our sample. To infer participation costs, we use 2, 423 auctions where some bidders where present and no one bid for the last units, *i.e.*, auctions similar to the one in Figure 12. Our interpretation is that the utility for all bidders is smaller than the participation costs, conditional on covariates. We use this information to partially identify participation costs (see page 24).

4.1 Set Up.

We use the three main specific features from the empirical setting to build our model. A SC is incurred only for the first unit bought while DMR are present for second to fourth units. The relative importance of the SC and DMR generate a trade-off, whereby bidders coordinate their behavior based on whether different units are complements or substitutes. A simple way to show this intuition is by assuming that the initial SC is proportional to the value of water, and DMR are linear in the number of units bought. We parametrize the SC effect by $(1 - \rho_1)$, whose interpretation is the percentage of water loss from the first unit because water is flowing through a dry channel (hence, proportional to the valuation of the bidder for

¹⁸A summary is available online in the online appendix at http://www.jdonna.org/water-auctions-web.

¹⁹Note that the results would be the same if participation costs were zero. However, the equilibrium when goods are complements would not be unique (see Subsection 4.2).

²⁰Weeks of January 22, February 5, April 5, May 1, May 8, May 15, May 22, May 29, June 5, June 12, July 3, July 10, November 26, and December 3.

the unit of water). One would expect that, conditional on rain, water loss would be constant within season with relatively more importance (lower ρ_1) in summer.²¹ We parametrize the DMR of unit k by ρ_k for k = 1, 2, ..., K. Let ρ be the vector of parameters that characterizes marginal utilities, *i.e.* $\rho \equiv (\rho_1, \rho_2, ..., \rho_K)$. Then, the marginal utility for bidder *i* for each unit k is:

$$MU_{ki} = \rho_k \cdot v_i,$$

where v_i , only known by bidder *i*, is a scalar that captures the valuation that the bidder assigns to their first (complete) unit of water, *i.e.*, when $\rho_1 = 0$ we have $MU_{1i} = v_i$.

We consider v_i to be independent and identically distributed on the interval \mathbb{R}_+ , according to the cumulative distribution function $F(v_i)$, for all bidders $i = 1, \ldots, N$. We assume that $F(v_i)$ admits a continuous density $f(v_i) > 0$ and has full support. It is assumed that $E[v_i] < \infty$. The distribution $F(v_i)$ is fully characterized by the parameter μ . The assumption that the support of $F(v_i)$ is bounded below by 0 is not restrictive, since bidders with negative valuations will not enter the auction. The private valuation, v_i , is only known by bidder i, and it is learned before entering the first auction.²²

The seller wants to allocate K identical units. These units are auctioned off sequentially by the seller using an English (ascending price) auction for every unit. All participating bidders observe the total number of individuals who take part of the auction, N. After every auction, each participant observes both the price paid by the winner and the winner's identity. The seller continues to run subsequent auctions sequentially until all the units are allocated. We assume that all bidders share the same utility function, $U(\cdot)$. The primitives of the model, (K, N, μ, ρ) , are common knowledge.

The strategy set for every bidder is the vector $\sigma \equiv (y_i^k, b_i^k)_{i=1,\dots,N}^{k=1,\dots,K}$, where $y_i^k \in \{0, 1\}$, $y_i^k = 1$ indicates that bidder *i* participates in the auction for unit k ($y_i^k = 0$ if bidder *i* does not participate in the auction for unit k), and b_i^k is the maximum amount that bidder *i* is willing to pay for unit *k*. Bidders play sequentially, or stage by stage. This means that they choose $\sigma_i^k = (y_i^k, b_i^k)$ after learning the outcome of the previous (k - 1) auctions. Bidders participating in auction *k* observe the price at which each bidder is no longer active (bids are observable) except for the winning bid. The information transmission is consistent with the auction being an English (or ascending price) auction rather than a second price auction.²³

²¹We discuss variation of SC across auctions (conditional on covariates) in page 21 in the paper.

 $^{^{22}}$ We do not consider the case where farmers might have different valuations for different units of water. The reason for this is that the units are identical and we condition on observables that may affect the price of water in the econometric specification (see Section 6.1). We obtained similar results by allowing the valuation for subsequent units to be different draws from the same distribution. The exposition of the model, however, becomes more cumbersome.

²³We model the game as in a button auction. Each bidder holds a button while the price continuously rises. A bid for bidder i is the value at which bidder i stops holding the button. When there are only two bidders active (holding the button) and one of them releases the button, the auction ends. The active bidder wins the object and pays the price at which the runner up stopped. See Cassady (1967) and Milgrom and Weber (1982) for details.

The seller allocates the unit to the highest bidder: $x_j^k \in \{0,1\}$ and $x_j^k = 1$ when $j = argmax(b_i^k)$ (and 0 otherwise), at a price equal to the second highest bid: $p^k = b_l^k$, where

 $l = \operatorname{argmax}_{i \neq j} (b_i^k)$. Let $X_i^k \equiv \sum_{j=1}^{j=k} x_i^j$ be the number of units that bidder *i* has won before participating in auction *k*. If only one bidder participates in a specific auction this bidder obtains the object for free. Each object is either allocated to one of the *N* bidders, or it is lost if none of the bidders decide to participate in the auction.

Participation decisions in each auction are done simultaneously by all bidders. To take part in every auction each bidder incurs a participation cost, c > 0, at the beginning of the period. We assume that participation in the first auction is free.²⁴ If only one bidder participates, this bidder obtains the object for free but he bears the participation cost, c, nonetheless. The process is then repeated in every period.²⁵

As discussed in previous subsection, the assumption of positive entry costs is consistent with the data in our empirical setting, where we observe no demand for some of the units, even though the reservation price is zero. The interpretation is that, in those situations where no-demand is observed, the value that bidders assign to that unit is smaller than the participation cost, $c.^{26}$

The utility for a bidder who buys l units and participates in m auctions is:

$$U_i(l, m, v_i; \rho, c) = \sum_{k=1}^{l} \rho_k \cdot v_i - \sum_{k=2}^{K} y_i^k \cdot c = \sum_{k=1}^{l} \rho_k \cdot v_i - (m-1) \cdot c$$

In the remainder of the paper we refer to $v_{N:N}$ as the highest realization of the random variables v_1, \ldots, v_n drawn independently from CDFs F_1, \ldots, F_N (one draw from each distribution), and $v_{N-1:N}$, as the second highest realization. More generally, $v_{j:N}$ is the j^{th} order statistic for a sample of size N from the distribution $F(v_i)$.

4.2 Four-Unit Auctions.

The most comprehensive independent unit of analysis that could be considered is the weekly auctions, encompassing all 40 units sold per week. This would be the relevant definition to answer questions related to demand fluctuations generated by supply shocks, such as no auctions due to drought or excessive rain, on an aggregate level. Alternatively, the narrowest

²⁴Allowing for a positive cost in the first auction will not affect the results qualitatively. However, the estimated distribution then will not be the original distribution of valuations $F(v_i)$ but the distribution of valuations conditional on v_i being greater than some minimum threshold for entering the first auction v^* , i.e., $F^*(v_i) \equiv F(v_i|v_i > v^*)$. Alternatively, we could impose some restrictions on the distribution of valuations to ensure that all bidders enter the first auction as in von der Fehr (1994).

 $^{^{25}}$ Bidders enter the auction if, and only if, the expected utility they obtain from the game is positive. See von der Fehr (1994) for a discussion of entry when the goods are complements or the conditions needed for entry when the entry cost in the first auction is positive.

 $^{^{26}}$ We later use this information to partially identify participation costs (see page 24). See above in this section for justification of this assumption in our specific empirical setting.

possible unit purchased is a *cuarta* (1 of the 40 weekly units). As discussed above in this section, the presence of SC and DMR indicate that *cuartas* within a day-schedule are not conditional-independent. Moreover, they are not the relevant unit of analysis to investigate individual farmers' demand, nor the price pattern described above.

Our original question is motivated by the price behavior caused by the *deterrence* effect. This particular behavior is observed within four-unit auctions and is the relevant unit of analysis in the model. This is an implication of the way the auction is structured: twelve hours of water (subdivided into four *cuartas* of three hours each) during day-time and twelve hours of water during night-time, each weekday. The logic behind this structure is related to water requirements in the area. First, water scarcity in the region made water accountability crucial. The standard unit used to measure surface area in Mula is called *tahúlla*. One *tahúlla* is, by definition, the surface area which can be irrigated in such a way that water level rises 1-foot high in 1 minute.²⁷ The surface area from one-*tahúlla* varies from one town to another, depending on soil conditions.²⁸ A four-consecutive units auction—half day, twelve hours of irrigation—is, in that sense, the amount of water that absorbed by a regular *parcela* (individual piece of land). Water requirements could and actually do differ (a) across farmers depending on past rainfall.

Second, the irrigation technique used in Mula is flood irrigation. The farmer builds small embankments in his *parcela* and water is delivered to the land by the channel system that simply flows over the ground through the crop. Flood irrigation requires a minimum of water delivery that, for a regular *parcela*, is captured by one *tahúlla*.

Finally, a supply-side consideration also plays a role. The reason to supply water for 12 hours (during day-time and during night-time) is to guarantee a particular and homogenous quantity for each *cuarta* (which depends on water pressure since water units are defined in hours). Given that the *De La Cierva* dam is continuously filled with water from the river, spreading the supply provision across weekdays ensures the homogeneity of water units.

Our data confirm these three points, validating the relevant unit of analysis for individual demand as four-consecutive units. The most frequent quantity purchased by farmers is twelve hours of water (42% of sold units are 4CU). There are no observations where the same farmer buys more than four consecutive units, nor observations where the same farmer buys consecutive units across days (*e.g.* there are no observations where the same farmer buys the last units of a day-auction, and the first units of the night-auction).

The next two assumptions allow us to determine the regime under which the game is being played.

 $^{^{27}}$ Although close in magnitude, the traditional Murcian measure of foot is not exactly the same as the foot measure used in the U.K. and the U.S. (Valiente 2001).

 $^{^{28}}$ The surface area of 1-tahúlla is 1,118 square meters in Murcia and 1,185 square meters in the old Kingdom of Aragón, except the region of Pías Fundaciones. The tahúlla has been used in regadío lands since the reign of Charles IV (king of Spain from 14 December 1788 until his abdication on 19 March 1808). In secano lands the surface area measure used is the fanega and the celemín. For further details see Vera Nicolás (2004).

Assumption 1 [A1]: $\rho_1 \leq \rho_4$.

Assumption 2 [A2]: $\rho_1 + \rho_2 \leq \rho_3 + \rho_4$.

When K = 4, we call it a strict complements regime when A1 and A2 holds. We call a weak substitutes regime when neither A1 nor A2 holds. The following results summarize equilibrium winning price behavior as a function of the model's primitives (valuations, SC, DMR, and participation costs). We later use these results for the estimation. We only consider pure strategy symmetric Perfect Bayesian Equilibrium (PBE).²⁹ All proofs and extensions are in Section B in the online appendix.

Proposition 1. In a strict complements regime (*i.e.*, when A1 and A2 hold) the pure strategy symmetric PBE is:

- First auction:
 - Participation: bidder i will always participate in the first auction, i.e. $y_i^1 = 1$.
 - Bidding Strategy: $b_i^1(v_i) = \sum_{k=1}^4 \rho_k \cdot v_i 3c.$
- Second, third, and fourth auctions:
 - Participation: bidder i participates in each auction if, and only if, she won the first auction, i.e. $y_i^k = 1$ if, and only if, $x_i^1 = 1$.
 - Bidding Strategy: If bidder i participates in each auction $(y_i^k = 1 \text{ for } k = 2, 3, 4)$, she will continue bidding until the price reaches its own valuation for that individual unit, $b_i^l(v_i) = \sum_{k=1}^4 \rho_k \cdot v_i - (4-l)c$.

Corollary 1. In a strict complements regime (i.e., when A1 and A2 hold) the total utility of the winner satisfies:

$$\sum_{k=1}^{4} p^k = \sum_{k=1}^{4} \rho_k \cdot v_{N-1:N} - 3c.$$
(1)

Lemma 1. In a weak substitutes regime (i.e., when neither A1 nor A2 holds) the probability that a bidder different from the winner enters the last auction is decreasing in the participation cost, c. Moreover, this probability goes to 1 when c goes to zero, i.e.:

$$\lim_{c \to 0} \left\{ \Pr\left(y_i^K = 1 \mid x_j^1 = 1, i \neq j\right) \right\} = 1.$$

²⁹When K = 2, cases where $\rho_2 \leq 0$ and $\rho_1 = \rho_2$ are equivalent to von der Fehr (1994), in Subsections 3.2 and 3.4, respectively. Uniqueness, however, is not proved by von der Fehr in any of those cases.

Corollary 2. In a weak substitutes regime (i.e., when neither A1 nor A2 holds) the marginal utility of the winner in the last auction, depending on how many units the winner won, satisfies:

If the winner won all four units:

$$p^4 = \rho_1 \cdot v_{N-1:N} - c. \tag{2}$$

If the winner won three units, two out of the first three, and the last one:

$$p^4 = \rho_2 \cdot v_{N-1:N} - c. \tag{3}$$

5 Regime Determination and End-Digit Preferences.

When goods are strict complements, very low prices—or, according to the auctioneer who ran the auctions, symbolic prices (Botía, Francisco, personal interview, Murcia, June 17, 2013)³⁰— are paid for the second, third, and fourth units by the winner of the first unit (Figure 2 and Table 1). This feature allows us to determine the regime (strict complements or weak substitutes) under which the game is being played using end-digit preferences and without specifying further assumptions on the model's primitives. When goods are strict complements a key prediction from proposition 1 is that the same bidder will win all units, pay his valuation for the whole bundle in the first auction, and pay a price of zero for the second, third, and fourth units. We do not observe zero prices (for second, third, and four units in the data), but very low prices (relative to the first price). These are symbolic prices. Although there is no reserve price in the actual auctions, we interpret the minimum price as a general agreement to bid a symbolic price in subsequent auctions. A common effect in our data is that farmers bid certain preferred end-digits prices substantially more often than others. We use this information to determine both regimes.

Studies of digit distribution go back to Benford (1938) who documented that in large data sets, leading digits are not distributed evenly (1 is the most common and 9 the rarest), and proposed a distribution for first digits of numbers in naturally occurring data. Abrantes-Metz, Villas-Boas, and Judge (2011) use Benford's second digit reference distribution to track the daily London Interbank Offered Rate (Libor) from 2005 to 2008 and find that in two periods, Libor rates depart significantly from the expected Benford reference distribution; collusion or rate manipulation appear as likely outcomes to this behavior, the authors suggest. Cramton and Schwartz (2000) also use end (trailing) digits to investigate collusive bidding in the spectrum auctions. Rauch, Goettsche, Braehler, and Engel (2011) use a Benford test to investigate the quality of macroeconomic data relevant to the deficit criteria reported to Eurostat by the European Union member states; they find that the data reported by Greece shows the greatest deviation from Benford's law among all euro states.

³⁰A summary is available online in the online appendix at http://www.jdonna.org/water-auctions-web.

In regards to end-digit preferences, Kandel, Sarig, and Wohl (2001) use Israeli IPO auctions to present evidence that investors have end-digit preference for round numbers (prices that end with 0 or 5), and that prices that end with 0 are used more often than those ending with 5.³¹ End-digit preferences and systematic age misreporting are important and broadly studied issues in demography, particularly in survey and census data when respondents inaccurately report ages or dates of birth (Myers 1940; Das Gupta 1975; Coale and Li 1991; and Siegel, Shryock, and Swanson 2003). The concern in these cases is, typically, heaping on particular ages such as those ending in 0 or 5. Crayen and Baten (2009) use some of these techniques to investigate the phenomenon of age heaping, and to test the hypothesis that an unequal distribution of human capital reduces welfare growth. Baker (1992) focuses on digit preferences in CPS unemployment duration data, where he raises the question of what can be said without making any specific assumptions concerning the true nature of end-digit preference in the CPS, and shows that employment duration is sensitive to the choice of a corrective for end-digit preference. Finally, end-digit preference has also been studied in the medical literature pertaining to individuals reporting body weight and height, blood pressure, and cigarette consumption (Bopp and Faeh 2008).

Table 3 shows in column 2 the frequency distribution by the last digit of price for first-unit prices.³² We observe strong preferences for 0 and 1, and somewhat weaker preferences for 5. In 32.1% of the cases we see a multiple of 10, in 32.2% we see a price ending in 1, while 8.4% report a multiple of 5. The frequencies also show some preference for 2 and 6, but not a marked one. After taking into account these effects, we find that 48% of the first-unit price observations are inconsistent with a uniform distribution in each digit.³³ That is, we would need to reclassify 48% of the cases to obtain a uniform distribution by digit. This is clearly not the case for our underlying distribution. We interpret these results as strong end-digit preference for 0 and weak end-digit preference for 5.

Strong preference for digit 1 is not, in general, an indication of a preference for this digit *per se* but, instead, a sign of competition. According to our model, first-unit prices are always competitive (in the sense that all N bidders will enter the auction when no information has yet been revealed), regardless of the regime. Nevertheless, second to fourth-unit prices are not competitive in the strict complements regime (competition in this regime takes part in the first unit where they bid for the whole bundle, and then pay a symbolic price for the second to fourth units since it is optimal for the remaining N - 1 bidders not to enter in these sequential auctions). Hence, end-digit preference for 1 in second to fourth-unit prices, as a sign of competition, are indicative of a weak substitutes regime. Alternatively, end-digit preference for 0 for second to fourth-unit prices are indicative of a strict complement regime.

³¹See Backus, Blake, and Tadelis (2013) for a recent application to negotiation in eBay auctions.

 $^{^{32}}$ We obtain similar patterns if we restrict the sample by month or schedule (day-time or night-time) or both.

³³This number corresponds to the value of the Whipple's concentration index (Siegel, Shryock, and Swanson 2003). In the absence of digit preference one would expect 10% in each terminal digit.

Moreover, in the strict complement regime the model predicts that all second, third, and fourth consecutive prices will simultaneously behave in this fashion. Column 3 in Table 3 display the frequency distribution by the last digit of price for the second to fourth units. Prices exhibit a pattern consistent with this description.

This behavior provides us with a natural lower bound for the strict complements regime, namely, second, third, and fourth unit prices within the same four-unit auction show a strong end-digit preference for 0. We use this behavior, along with the model, to identify the two regimes.³⁴

Figure 11 displays the histogram of the percentage change of first price against the median of second to fourth price, by regime.³⁵ It can be seen in the figure that end-digit preference behavior (as defined above) also captures, in general, the other empirical prediction from the model, namely, that prices are competitive in the weak substitutes regime but exhibit the deterrence effect in the strict complements one (the way to see this in the figure is that percentage change from the first to the second, third or fourth prices is high when goods are strict complements). This is remarkable as the end-digit preference behavior used to identify the regime is unrelated *a priori* to this second empirical prediction. This provides further evidence in favor of the model.³⁶ Regime identification is done by using the strongest version of the empirical prediction to identify the strict complements case, *i.e.*, the case in the left panel in Figure 11.

A final robustness check further shows that the approach in this section consistently identifies both regimes in terms of our model. Columns 1 and 2 (first unit) in Table 4 display, by regime, the frequency distribution in terms of end-digit prices for first-unit, among each four-unit auction. As emphasized above, both regimes should exhibit competition for first-unit prices according to our model. This competition is captured by the same distribution among ending digits in both regimes. This is what we observe in columns 1 and 2 (first unit). Columns 3 and 4 in Table 4 (fourth unit) show that, as predicted by the model, fourth-unit prices for weak substitutes are also competitive: 29.5% of preference for 0 vs. 39.7% for 1.

 $^{^{34}}$ We could also use weaker or stronger definition of end-digit or round-number preferences to obtain different bounds for the empirical distributions of prices in each regime. We could, for example, assume that in the strict complements regime second, third, and fourth unit prices within the same four-unit auction show simultaneously a strong end-digit preference for 0 or 5. Our results are robust to include end-digit preference for 5 as well.

Note that the strongest version of the empirical prediction is be that all second, third, and fourth prices display an end-digit preference for 0 in any given four-unit auction for the same individual. A weaker version would be that two out of the three (among second to fourth) prices show an end-digit preference for 0. The weakest version is that just one of these three prices exhibit an end-digit preference for 0. The last (weakest) specification only provides us an upper bound for strict complements regime identification since, as shown in Table 3, the underlying distribution displays an end-digit preference for 0 even in the weak substitutes regime.

 $^{^{35}}$ The figure looks similar if we use the second, or the third, or the fourth, or the average of second to fourth prices.

 $^{^{36}}$ Using a modified version of this assumption that differentiates end-digit preference for prices ending in 0 that exhibit more frequency (for example, prices like 100 are more frequent than 150) yields almost identical results.

Note that, in column 3 (strict complements), the percentage of observations with last digit 0 is 100% by construction.

6 Estimation.

6.1 Econometric Specification.

We estimate the model via maximum likelihood using an exponential distribution for the individual valuations. In this subsection we describe how the likelihood is formed and how we account for rain expectations and auction heterogeneity.

Regime Determination. When goods are strict complements, very low prices—or, according to the auctioneer who ran the auctions, symbolic prices (Botía, interview)— are paid, by the winner of the first unit, for the second, third, and fourth units (Figure 2). The predicted price pattern by our model for each each regime (strict complements and weak substitutes) provides us with a straightforward empirical method to determine them (see Section 5). This allows us to separate data into four categories:

a) Same bidder wins all four units and goods are in a strict complements regime (i.e. when A1 and A2 hold),

b) Same bidder wins all four units and goods are in a weak substitutes regime (i.e. when neither A1 nor A2 holds),

c) Last winner also bought two out of the first three units, three units in total, and goods are in a weak substitutes regime (i.e., when neither A1 nor A2 holds),

d) Otherwise.

Categories a, b, and c define the three mutually exclusive regions of the likelihood. In region a, winning prices are determined by equation 1. In region b, winning prices are determined by equation 2. In region c, winning prices are determined by equation 3. Let D^a be an indicator variable that equals 1 if the winning price is in region a, and 0 otherwise. Define analogously D^b , D^c for regions b, and c, respectively (so $D^a + D^b + D^c = 1$). See subsection 6.2 for a discussion about the regions of the likelihood and the covariates.

Identification. For the case of an English auction, the conditional distribution of private valuations is non-parametrically identified when the transaction price and the number of bidders are observable (Athey and Haile 2002). This result is immediately useful in our sequential English auction model where bids are conditional-independent draws from a distribution $F_V(.)$ and the equilibrium (observed) transaction price is a function of the second highest valuation, $v_{N-1:N}$. Consider the strict complements regime. Winning prices are determined by equation 1. The distribution of valuations is identified up to the multiplying constant, $\sum_{k=1}^{4} \rho_k$, using equation 1 and the result from Athey and Haile 2002. Identification

of the remaining parameters, ρ_k , would require four additional independent restrictions (in addition to equation 1). Two additional restrictions are provided by the model from corollary 2 (equations 2 and 3). But we only observe winning bids in the data. Then, two of the ρ_k , $k = 1, \ldots, 4$, are not identified without further structure. So we use a specification with linear decreasing marginal returns due to the mentioned data limitation. Linear decreasing marginal returns impose two additional restrictions. First, we define $\rho_1 = 1 - \alpha$ and $\rho_2 = 1 - \beta$. (Note that these are not restrictions on the parameter space.) We then restrict the parameter space by assuming that $\rho_3 = 1 - 2\beta$ and $\rho_4 = 1 - 3\beta$ (*i.e.* linear decreasing marginal returns). Hence, we have three independent restrictions (equations 1, 2, and 3) and three parameters to estimate (μ, α, β), where μ is a parameter that fully characterizes the distribution of valuations.³⁷ With observability of all bids (not just the winning bids as in our empirical setting), ρ_3 and ρ_4 would be identified and we would not need to impose the linearity assumption on marginal returns.³⁸

Farmers' Expectations of Future Rain. For our estimation we allow DMR, β_t , to vary across auctions holding fixed SC, α (more about this below). We allow β_t to vary with farmers' expectations of rain in each auction $t = 1, \ldots, T$. We proxy these expectations by actual (*i.e.* observed) future rain, so $\beta_t = \beta_0 + \beta_1 R_t^F$, where R_t^F is a dummy variable (defined next) that is linked to expectations about future rain in t, and β_0 and β_1 are parameters. $R_t^F = 1$ if farmers expect that rain is going to be positive (for the day for which they are buying water) and zero otherwise. We further let β_t have different intercepts in each regime:

$$\beta_t^S = \beta_0^S + \beta_1^S R_t^F
\beta_t^C = \beta_0^C + \beta_1^C R_t^F .$$
(4)

Table 6 provides an heuristic argument to understand the reasons behind this equation. The table presents probit regressions of a dummy variable identifying the regime (strict complements vs. weak substitutes) on future rain and other covariates. We interpret future rain in these regressions as a proxy for aggregate expected future rain for the farmers. Table 6 shows that low expected rain and high demand months (May to August) significantly increase the likelihood of being in a strict complements regime. The interpretation is that farmers have some information (expectations) about future rain. While the idiosyncratic component of this information is captured by their type, v_i , the common component is captured by β_t . When farmers expect, on aggregate, no rain in a given day, they will coordinate to play in the strict complements regime. Seasonality also affects the demand for water and affects the position of a farmer in the production curve (Figure 10). The results in Table 6 show that it is the slope on the marginal return effect that drives the change of regime, holding fixed SC.

In our parametrization we fix α across auctions and season but we allow β_t to vary. We

 $^{^{37}}$ Note, however, that the distribution of private valuations is non-parametrically identified from the result from Athey and Haile 2002.

³⁸Note that assumptions A1 and A2 are equivalent to assume that $\alpha \ge 4\beta$.

expect α to vary across auctions and seasons as well. But this variation is not separately identified from the variation on β_t because it is the relative magnitude of the effects that matters. The rationale for why we let β_t vary (instead of α) is that a regime switch is driven by the (residual) demand for water by the farmers, as determined by rain and seasonal effects. Therefore, the estimated changes in β_t should be interpreted relative to changes with respect to α .³⁹

The Likelihood. The econometric problem consists of finding the parameter that characterizes the common distribution of valuations F and the structural parameters that best rationalize the bidding data. As discussed in the previous section, the bid levels at which bidders drop out of the auctions are not observed, except the bidder with the second-highest valuation. We estimate the model *via* maximum likelihood assuming that farmers draw independent and private valuations from an exponential distribution at each four-unit auction, conditional on observed auction-specific covariates. (We discuss the assumptions below in Subsection 6.2.)

Our model and the context of the market under analysis provide insight on how the characteristics of farmers and auctions should affect private values, but it offers little guidance on the functional form of this distribution. We assume that farmers' valuations, v_i , follow an an exponential distribution for each four-unit auction.⁴⁰ In Subsection 7 we report the results from a Kolmogorov-Smirnov test where the null hypothesis that the distribution of private valuations are draws from an exponential distribution cannot be rejected.

Let $v_i \sim F(v;\mu)$, where $F(v;\mu) = (1 - e^{-\mu v}) \mathbf{1} \{v \ge 0\}$ is the CDF of an exponential distribution that is characterized by the scalar $\mu > 0$. Equations 1, 2, 3, and 4 jointly identify the parameter vector $(\mu, \alpha, \beta_0^C, \beta_0^S, \beta_1^C, \beta_1^S)$, conditional on the regime (see Section 5) and exogenous covariates, R_t^F . The full system of equations is given by:⁴¹

³⁹We obtained similar results to the ones on Tables 7 and 8 fixing β and allowing α_t to vary in each auction $t = 1, \ldots, T$.

 $^{^{40}}$ In our earlier working paper Donna and Espin-Sanchez 2012 we used an Exponentiated Gamma (EG) distribution The EG distribution gives us a closed-form solution for the PDF of the j^{th} order statistic and is characterized by a single parameter. Additionally, the PDF of the j^{th} order statistic of a EG is a weighted average of several PDF of EG. This implies that the PDF of any order statistic of an EG distribution also has a closed-form solution.

⁴¹The third equation in the system is, actually, $p_b^4 = Max \{(1 - \alpha)v_{N-1:N}, (1 - \beta)v_{N-2:N}\}$, since we do not know whether the runner-up in the last auction was the bidder who already won one unit or a bidder without previous purchases. However, when N is large, $(1 - \alpha)v_{N-1:N} < (1 - \beta)v_{N-2:N}$ if $\beta \simeq \alpha$. But, in the case that $\beta \simeq \alpha$, the same bidder will not win three out of four units. That is, in an auction where N is large and the same bidder wins three out of four units, we expect β to be significantly greater than α . Therefore, the equation can be simplified to $p_b^4 = (1 - \alpha)v_{N-1:N}$.

$$\sum_{k=1}^{4} p_{a}^{k} = \left[4 - \alpha - 6\beta_{t}^{C}\right] v_{N-1:N} - 3c$$

$$p_{b}^{4} = (1 - \alpha)v_{N-1:N} - c$$

$$p_{c}^{4} = (1 - \beta_{t}^{S})v_{N-1:N} - c$$

$$\beta_{t}^{S} = \beta_{0}^{S} + \beta_{1}^{S}R_{t}^{F}$$

$$\beta_{t}^{C} = \beta_{0}^{C} + \beta_{1}^{C}R_{t}^{F}.$$
(5)

Let $\theta \equiv (\alpha, \beta_0^C, \beta_0^S, \beta_1^C, \beta_1^S)$ and let v_i be a conditional-independent draw from $F(\cdot; \mu | \theta, R_t^F)$. Then, the likelihood function is given by:

$$L(\cdot; \mu \mid \theta, R_t^F) = \prod_{t=1}^T f_{N-1:N} \left(\frac{\sum_{k=1}^4 p_t^k}{4 - \alpha - 6(\beta_0^C + \beta_1^C R_t^F)}; \mu \mid \theta, R_t^F \right)^{D_t^a} \times f_{N-1:N} \left(\frac{p_t^4}{1 - \alpha}; \mu \mid \theta, R_t^F \right)^{D_t^b} \times f_{N-1:N} \left(\frac{p_t^4}{1 - \beta_0^S + \beta_S^C R_t^F}; \mu \mid \theta, R_t^F \right)^{D_t^c},$$
(6)

where $f_{N-1:N}(v;\mu)$ is the probability density function (PDF) of the $(N-1)^{th}$ order statistic from a sample of N from the exponential distribution of valuations F, $D_t^a + D_t^b + D_t^c = 1$ $\forall t$, and D_t^a , D_t^b , D_t^c are, respectively, indicator variables for cases a, b, and c, as defined above at the beginning of this subsection .

Auction Heterogeneity. We allow the mean of the distribution of valuations to depend on various characteristics that are drawn from the data. We assume that observed prices follow a linear function of the following exogenous variables and estimate all parameters using the likelihood function:⁴²

$$\mathbb{E}\left(v_{t}^{i}\right) = Z_{t}^{\prime}\gamma = \gamma_{0} + \gamma_{1}R_{t}^{P} + \gamma_{2}\left(R_{t}^{P}\right)^{2} + \gamma_{3}Night_{t} + \sum_{k=2}^{5}\gamma_{2+k}Day_{t}^{k} + \sum_{k=2}^{12}\gamma_{6+k}Month_{t}^{k}.$$
 (7)

The first exogenous variable, R_t^P , refers to *Past Rain*, a moving average of the daily rain beginning seven days prior to the date of the auction; we include a quadratic term to allow for non linearities in past rain. The second variable is a dummy variable that equals one if the water was bought for night use. The next four variables are a set of dummy variables for each weekday. Finally, the last eleven variables are a complete set of monthly dummy variables to condition on seasonality. Water prices soar in this market during the dry summer and drop in

 $^{^{42}}$ Laffont, Ossard, and Vuong (1995) assume that private values follow a log-normal distribution and let the mean of the logarithm of the valuations be a linear function of exogenous characteristics. Haile and Tamer (2003) condition on covariates by constructing the conditional empirical distribution functions using Gaussian kernels. See Hickman, Hubbard, and Saglam (2011) for a recent guide to the literature on structural econometric methods in auctions.

winter. We accommodate these shocks to demand with seasonal monthly dummy variables. See Sections D and E in the online appendix for details about the estimation procedure.

Identification and Estimation of Participation Costs. Although, throughout the previous estimation procedure, participation costs, *c*, have been fixed at an arbitrary small magnitude, we recover them from our data. We use our model and data where auctions were run, no bids observed and farmers were present, along with the structural estimates. Participation costs are identified by the necessary condition for a bidder to bid in the first auction that is given by:

$$(1 - \alpha)v_{N:N} < c.$$

More generally, a condition that additionally involves second, third, and fourth marginal utilities for the case where the bidder also enters the individual auctions for two, three or four units should be considered. In these cases, participation costs are also greater than the average marginal utility for second, third, and fourth units. Formally:

$$Max\left\{(1-\alpha), \frac{(2-\alpha-\beta_t)}{2}, \frac{(3-\alpha-2\beta_t)}{3}, \frac{(4-\alpha-3\beta_t)}{4}\right\} v_{N:N} < c.$$
 (8)

Note that, when $\alpha < \beta_t$, the former condition is sufficient, implying the latter. In our econometric specification the structural parameter α is fixed while the parameter β_t varies according to the farmers' expectations of (exogenous) future rain. One would expect to observe auctions without bids when farmers' expectations for rain, as captured by actual future rain, are high (which in the model is represented by a relatively high β_t). Therefore, absence of bids will only occur when $\alpha < \beta_t$, thus, the former identification restriction is sufficient.

Analogously, using the model and the remaining data not used in the structural estimation, we obtain an upper bound using that participation cost are lower than the minimum registered price (conditional on covariates, sunk cost, and decreasing marginal returns).

6.2 Discussion.

Conditional Independent Private Valuations (CIPV). For the estimation we assume that farmers have independent and private valuations at each four-unit auction, conditional on observed auction-specific covariates. The first justification for CIPV is that each bidding farmer (who may or may not be a water-owner) has his own land extension, and his own mixture of trees and crops. This eliminates a strict common value scenario. In addition, in the econometric specification we account for observables that affect all farmers in a similar way such as (past and future) rainfall, schedule of the auction, day of the week, weather seasonality, etc. (see subsection 6.1 for details). Second, the products sold are units of water. Assuming that farmers have private information from other farmers about the characteristics

of this product is not in line with the homogeneous nature of water units. Finally, the conditional-independence assumption is the most credible in our context, given the varying nature of farming products and soil conditions across farmers. To understand why, recall that sellers in the water market are a holding formed by the water owners and buyers are farmers that own fertile land. Around 500 different farmers are observed to win auctions in our sample. Not all of these farmers show up at every auction or decide to participate if they are present. Farming products cultivated in the area are mainly fruit and citrus trees (lemon, orange, peach, mandarin, and apricot), and vegetables (tomato, lettuce, and onion). The amount of water required by the trees depends on the time of the year and type of crop (citrus trees should not be irrigated daily). Moreover, and given that we condition on seasonality, water requirements vary across products. For example, water needs for grapefruit and lemons are about 20% higher than those for oranges, while water requirements for mandarins are about 10% less. Ground conditions (which also vary across areas where different farmers have their land) also affect water necessity.⁴³ The variations across farmers generated by these factors provide support for the fact that the conditional-independence assumption seems satisfied, given that each day the market is quite specific and since we work with data for four-consecutive auctions as a unit of analysis (sequential auctions).⁴⁴

Auction Heterogeneity. Observed heterogeneity across auctions arises due to seasonal effects, rain, and the day and time of the week when the auction occurs. This means that the distribution of private values for the t^{th} auction, $F_t(\cdot)$ is not constant across auctions. In our estimation, we recover the family of distributions $F(\cdot|Z_t, \gamma)$. That is, we assume for every four-unit auction that $F_t(\cdot) = F(\cdot|Z_t, \gamma)$, where $\gamma \in \mathbb{R}^k$ is a parameter vector and Z_t is a vector of fully observed characteristics describing the environment of the t^{th} auction. We described the inclusion of these covariates above.

Number of Potential Bidders. The number of potential bidders in each auction, N_t , is not observed. Moreover, it is not identified (Athey and Haile 2002). We assume that it is constant for every four-unit auction, $N_t = N$. Table 5 displays the timing structure for different bidders in our sample. For our estimation, we let the number of potential bidders in each auction be the yearly average of different farmers who won auctions in our sample.⁴⁵

⁴³Table A3 in the online appendix displays appropriate intervals for watering citrus.

⁴⁴Our justification of the CIPV paradigm is in line with the literature on empirical auctions. For first price descending auctions see, for example, Laffont, Ossard, and Vuong (1995) in an application to agricultural products (greenhouse eggplants in Marmande, France) where the number of bidders vary between 11 and 18. For English auctions, Haile and Tamer (2003) apply their limited structure model to U.S. Forest Service timber auctions, where the number of bidders vary from 2 to 12.

⁴⁵The agricultural products that are cultivated in the area are mainly citrus trees, which are harvested once per year. The number of different bidders who bought at least one unit during a specific year constitutes a good approximation of the number of farmers who were actively bidding in each four-unit auction during that year. The monthly average of different bidders who bought water in the sample (years 1954 to 1966) is 8.31 (Table 5).

We estimate the model using different values of N for robustness.⁴⁶

Unobserved Heterogeneity. Throughout, we have assumed that the vector Z_t of covariates is fully observed by the econometrician. In our environment, unobserved heterogeneity implies that the distribution of bids may not be conditional-independent across t. All farmers may, for example, observe some factor unobservable by the researcher that shifts the location of the distribution values. This unobserved heterogeneity could lead to correlation among bidders' valuations, causing an identification problem and inconsistent estimates to arise.⁴⁷ Modeling unobserved heterogeneity may require additional assumptions on the behavior of unobservables, such as independence, separability, strict monotonicity, and is beyond the scope of this paper.⁴⁸

Dynamic Strategic Considerations. The way in which the auction system is carried out every week raises the question of the importance of dynamic strategic considerations between four-unit auctions both among days (Monday to Friday for a specific schedule) and between schedules (day-time vs night-time for a specific day). Tables 1 and 2 show that winning prices decline across days (for a given schedule) and at night (for a given day), which is consistent with the literature on empirical sequential auctions. These dynamic strategic considerations are outside the scope of the present investigation, and we abstract from them in the model.⁴⁹ However, it is important to note that, even if present, dynamic behavior considerations do not invalidate the model's assumptions. As emphasized above, the conditional-independent units of analysis are four-unit auctions (not day-auctions of eight units or week-auctions of 40 units) which, conditional on covariates, are homogeneous goods. As can be seen from the correlations presented in Table 2, previous patterns are consistent along the whole sample and robust to the inclusion of a whole set of fixed effects and covariates. The principal difference between prices in these four-unit auctions is related to the uncertainty of future rain. As it is explained above, we include covariates for schedule, day-of-the-week, and past rain in our structural estimation that capture technological or strategic effects. Future rain, on the other hand, is also included as a proxy for farmers'

⁴⁶In Table 7 we present the results for $N \in \{8, 10\}$. We have performed a sensitivity analysis to different values of N_t that are consistent with the pattern observed in Table 5 and the evidence described in Section 3. In addition, we broke the sample into four periods and performed the estimation independently in each period allowing the mean value of N_t to vary by period. We obtained similar results to the ones reported in Table 7.

⁴⁷From the agricultural census data we observe individual characteristics of the farmers which we are able to link to the winning bids. Given the structure of the agricultural water market we are modeling, it does not appear to be an important concern once we consider the homogeneity of the selling good and the observed characteristics we introduce in our estimations (seasonality, past and future rain, among others).

⁴⁸For a discussion on this issue see, among others, Athey, Levin, and Seira (2011) for an application to timber auctions, and Krasnokutskaya (2011) for a semi-parametric approach to Michigan highway procurement contracts. Roberts (2009) uses information contained in reserve prices to allow bidders' private signals to depend on the realization of the unobserved heterogeneity. Balat (2013) allows for unobserved heterogeneity using dynamic auctions in the highway procurement market.

⁴⁹For a broader discussion see Donna and Espin-Sanchez 2013a.

beliefs to account for these possible strategic behaviors unaccounted by previous covariates. In that sense, our estimates should be interpreted as four-unit day-schedule specific auctions, conditional on past rain and seasonality. It seems implausible that after accounting for these observables and unobservables,⁵⁰ and given that the relevant unit of analysis is the four-unit auction, dynamic behavior would affect our results concerning individual demand.⁵¹

Regions of the Likelihood and Covariates. Another concern may be selection in the regions of the likelihood. As emphasized in subsection 6.1, categories a, b, and c define the three mutually exclusive regions of the likelihood. (In region a, winning prices are determined by equation 1. In region b, winning prices are determined by equation 2. In region c, winning prices are determined by equation 3.) Table A4 in the online appendix displays a comparison of the covariates in the three regions of the likelihood. As expected, prices are higher in the strict complements regime (region a) relative to the weak substitutes regime (regions b and c). This is because the amount of rainfall is lower under the strict complements regime (region a) relative to the weak substitutes regime (regions b and c). Rainfall is lower in region a) (relative to regions b and c) due to weather seasonalities: the percentage of observations in Apr-May (when the agricultural products need the water the most) is substantially higher in region a) (strict complements) relative to regions b) and c) (weak substitutes). The opposite is true during the low demand season (Jan-Mar and Oct-Dec). Finally, note that there is no substantial variation (between the strict complement and weak substitutes regimes) in terms of the percentage of observations by Schedule (day or night) and Weekday (Mo, Tu, We, Th, and Fr).

7 Results.

7.1 Maximum Likelihood Estimates.

In this section we present the estimation results under various econometric specifications. We present the structural estimates obtained using a tolerance level of 1.0e - 12. We let private valuations for each four-unit auction follow an exponential distribution, and follow the described estimation procedure. As discussed above, the number of bidders, N, is determined by the monthly average of different bidders who bought water in the sample (years 1954 to 1966). In this 13-year sample, the average is slightly above 8. Each of these farmers regularly won auctions. It is reasonable to assume that they attended the auctions. Tables 7 and 8 present our estimation results. Columns 1, 3, and 5 present the estimates for N = 8, while

 $^{^{50}}$ While farmers use their reasonable good predictions in their decisions, we use actual future rainfall in our estimation.

 $^{^{51}}$ Once we condition on these covariates, the concern that a bidder's outside option would vary according to the day of the week (or schedule) is addressed by redefining the idiosyncratic individual valuation in such a way that the new one be the original valuation net of the outside option. By normalizing the outside option of Friday-night to zero the model's assumptions remain valid.

columns 2, 4, and 6 do it for $N = 10.5^{2}$ For each specification, we present the estimates of the model's structural parameters in Table 7 and the estimates of the covariates in Table 8. Table 8 is the continuation of Table 7. That is, for each specification (column) in Table 7, Table 8 displays the estimates of the covariates in that specification.

All parameters have the expected signs. We use the estimate of the parameter γ (that characterizes the distribution of private valuations), to compute the mean valuation of the first complete unit of water. In the case of column 3, the value of the first complete unit of water is 152.93 *ptas*. As expected, in the specification in column 4 (with 10 different bidders), the mean value of the first complete unit of water is slightly lower, 138.9 *ptas*.

The parameter β_1^R , $R \in \{C, S\}$ captures the effect of future rain. As farmers' expectations of future rain increase, DMR are more severe $(\beta_1^R > 0, R \in \{C, S\})$. This increases farmers' likelihood of coordinating in a not-strict complements regime (see Table 6) and thus reduces their valuation of subsequent units of water $(\frac{\partial p_t^i}{\partial R_t^F} < 0)$. Predicted DMR are obtained by adding the estimates of intercepts, $\hat{\beta}_0^R$, $R \in \{C, S\}$, to the estimates of the slope, $\hat{\beta}_1^R$, $R \in$ $\{C, S\}$, conditional on the rain on the day of the auction. When evaluated at the average future rain from each regime, the following null hypothesis (joint test) that overall DMR are lower in the strict complements regime (as predicted by the model) cannot be rejected (pvalue above 10%). $H_0: \quad \hat{\beta}_0^S + \hat{\beta}_1^S \hat{\mathbb{E}}_s(R_t^F) > \hat{\beta}_0^C + \hat{\beta}_1^C \hat{\mathbb{E}}_c(R_t^F)$, where $\hat{\mathbb{E}}_s(R_t^F) = \frac{1}{T_s} \sum_{t:D_t^a=0} R_t^F$, $\hat{\mathbb{E}}_c(R_t^F) = \frac{1}{T_c} \sum_{t:D_t^a=1} R_t^F$, T_s , and T_c are the number of auctions in not-strict complements and strict complements regimes, respectively.

The estimates of the SC parameter, α , are statistically significant in all specifications. Given the choice of parametrization for sunk costs, the parameter estimates can be interpreted as the percentage loss in terms of a complete unit of water (Section 4). For our estimate in column 3 this represents a loss of 4.6 *ptas* (using the mean value of 152.9 *ptas* for a complete unit).

The estimated coefficients for covariates have the expected sign. For specification 3, for instance, prices in August (February) are significantly 234 *ptas* higher (11 *ptas* lower) than on January. This is consistent with the conventional wisdom that water is more (less) valuable during these months because of high (low) water demand. Also as expected, past rain decreases observed prices in the data. For specification in column 3, an increase in the average rainfall by 1 mm from the previous week (with respect to the day of irrigation), decreases average conditional price of a unit of water by 1.7 *ptas*.

⁵²In their simulated Non Linear Least Squares (NLLS) estimation, Laffont, Ossard, and Vuong (1995) search for the best value of N by minimizing a lack-of-fit criterion (proposition 4). Note that, as discussed in Subsection 6.1, identification of the distribution of valuations and structural parameters of our model requires observation of the total number of bidders. The rationale for this is straightforward: whether second highest realization of the random variable v_i is from a sample of size N = 10, or from a sample of size N = 100, it is crucial to interpret the second highest bid (observable in our data). Although observation of an additional order statistic can eliminate this requirement (Song 2004), this would require imposing further structure on the distribution of beliefs in our model (to interpret auctions where, for example, three different farmers win auctions), which is outside the scope of this investigation. Moreover, we only observe winning bids in the data (see Section 3).

Participation cost are recovered using data where auctions were run with farmers present, but no bids were placed, along with the identifying restriction that holds in such cases.⁵³ Out of the 3, 203 auctions where no bids were placed, we use the 2, 423 where some bidders where present (auctions similar to the one in Figure 12). We obtain the following interval estimate using specification 3: $0.0082 < \hat{c} < 0.1431$. That is, participation costs are positive but small (less than 14 cents of a *peseta*). This is in line with the intuition from the model: hassle or opportunity costs because farmers value their time.

7.2 Discussion.

Robustness and Goodness of the Fit. In comparing columns 1-2 and 3-4, it is clear that the model with covariates outperforms the model without, as shown by the significance of past rain and seasonal dummy estimates, the increase in the likelihood function, and the improvement in the goodness of the fit. The main reason is the dependence of prices on seasonal factors, which we capture in our specification with seasonal dummy variables. From the residual analysis we find no evidence that the increase in the log likelihood function is due to the parametric misspecification of the value distribution itself. Our specification survives the Kolmogorov-Smirnov test, so that the exponential distribution of private valuations cannot be rejected (for the specification in column 3 the *p*-value of the test is 39%).⁵⁴

As regards the goodness-of-fit, our specification in column 3 performs quite well. The pseudo $-R^2 = 53\%$ is obtained by computing predicted prices by our model: $pseudo - R^2 = 1 - \frac{\sum_{t=1}^{T} (p_t - \hat{p}_t)^2}{\sum_{t=1}^{T} (p_t - \hat{p})^2}$, where \hat{p}_t are prices predicted by the model and \bar{p} is the mean of prices. These results are in line with the R^2 obtained in the reduced-form regressions. Although not directly comparable given the distribution assumptions in the structural approach, the $R^2 = 23\%$ in the reduced-form specification with all covariates (column 3 in Table 2) can be heuristically interpreted as the proportion of variability in the data set that is accounted for by the covariates. The proportion accounted for by the model without covariates displayed in column 1 in Table 7 is $R^2 = 28\%$.⁵⁵ As can be seen in Figure 13, our model allows us to follow winning prices accurately.⁵⁶ The figure displays real prices against predicted prices using three different models: (i) our structural model (specification 3 in Table 7), (ii) a standard English auction model (specification 5 in Table 7, that we discuss in the next subsection), and (iii) a reduced-form model (specification 4 from Table 2 that includes as regressors *Past Rain* and multiple fixed effects, including individual fixed effects).

 $^{^{53}}$ See page 24.

 $^{^{54}}$ We perform the nonparametric test to evaluate the equality of two distributions of valuations: our sample of private values with a reference from an exponential distribution.

⁵⁵If we additionally add individual fixed effects to the reduced-form specification, the R^2 just increases from 23% to 36% (column 4 in Table 2).

 $^{^{56}}$ We describe how we compute the predicted prices in section C in the online appendix. See also section C in the online appendix for a high definition version of this figure.

Understanding the Importance of the Model. We proceed now to analyze our model's implications with respect to the importance of SC and DMR. Suppose that the researcher neglects the dynamics that arise from the model and, instead, estimates a standard English auction model. Suppose, for instance, that we are in the strict complements regime and that valuations follow a distribution with mean, μ_v , and standard deviation, σ_v . Then, the estimated mean of the distribution of valuations using the standard model will be underestimated: $\mathbb{E}(\hat{v}_i)^{SM} < \mathbb{E}(v_i) = \mu_v$, where SM stands for standard model. Similarly, the estimation of the standard deviation of valuations will be overestimated: $\mathbb{V}(\hat{v}_i)^{SM} > \mathbb{V}(v_i) = \sigma_v$.⁵⁷ The same is true in the weak substitutes case.

Overestimation of the variance of the distribution is caused by attributing the variation in prices (among different units) to a relatively more dispersed underlying distribution. The farmer actually pays for the whole bundle in the first unit, thus deterring the entrance of other bidders in the remaining three auctions. The mean is underestimated when the common SC and DMR are not accounted for in the estimation. In the case of the exponential distribution used in our specifications, this failure translates into an underestimation of the parameter μ .

Columns 5 and 6 in Table 7 present the estimates from a standard English (button) auction. Aside from the mentioned bias in the parameter that characterizes the distribution, the results in these columns indicate that taking SC and DMR into account significantly contributes to the model's explanatory power. Figure 13 shows predicted prices from the standard (button) English auction model (specification 5 in Table 7), and compares them with actual prices and with those from our structural model (specification 3 in Table 7).⁵⁸ Consistent with these results, the *p*-value for the null hypothesis that $\hat{\alpha} = \hat{\beta}_0^C = \hat{\beta}_0^S = \hat{\beta}_1^C = \hat{\beta}_1^S = \hat{c} = 0$ is less than 10^{-4} .

An alternative approach is to ask how the incomplete model from Haile and Tamer (2003) can be adapted to the present case.⁵⁹ This alternative approach relies on two basic assumptions with intuitive appeal: (i) bidders do not bid more than they are willing to pay for a unit, and (ii) bidders do not allow an opponent to win at a price they are willing to beat. In our case, with SC and DMR, these two simple assumptions are violated. In the strict complements regime, bidders bid according to $b_i^1(v_i) = [4 - \alpha - 6\beta]v_i - 4c > v_i$, violating (i), and no bidder (except the highest type) participates in the second to fourth unit auctions, violating (ii). In the non weak substitutes regime, both assumptions are also violated, though the intuition is different. In this case, the equilibrium is only partially revealing: bidders' strategies are step functions, so the equilibrium is semi-pooling. When α is greater but close to β , bidders bid above their valuations to intimidate other bidders and deter entry

⁵⁷In the strict complements case, and given a fixed number of potential bidders N, the (true) mean and variance of the N-1 order statistic will be greater than the estimated using the standard model because the (true) price paid will be $[4 - \alpha - 6\beta_{t,c}] v_{N-1:N} - 4c$ and not $4v_{N-1:N}$ (predicted by the standard model). ⁵⁸See footnote 56.

⁵⁹Larsen (2013) uses a similar approach to Haile and Tamer to obtain bounds about the primitives in an auction model followed by dynamic bargaining with two-sided incomplete information without solving for the equilibrium of the game.

in the second auction, thus (i) is violated. Additionally, the same argument as in Black and De Meza (1992) and Liu (2011) applies when goods are substitutes. The winner of the first auction imposes a negative externality on himself. His willingness to pay for the second unit is lower than it was for the first unit, making him a weaker bidder in such situations. Given that all bidders will internalize this effect, some will bid below their marginal utility for the object in the first auction. The greater are DMR, β , the greater this underbidding effect will be.

Applying these assumptions to the four-units bundle would not to produce informative bounds because marginal valuations of the units differ according to the regime and the number of different winners per four-unit auction. Bundling the four-unit or even applying Haile and Tamer's approach separately for each regime, requires the model in Section 4 as an interpretation of the underlying behavior.⁶⁰

Complementarities are not Collusion. An alternative hypothesis of farmers' behavior in the strict complements regime is that bidders might be playing some collusive (noncompetitive) strategy. As emphasized in Section 3, the demand side of this market for water is composed of as many as hundreds of farmers (Table 5). Even when farmers attend the auction and do not bid, the observed number of different winners is relatively high (Figure 12). (Note that all auctions were run in weeks similar to the one in Figure 12, so water for units 17-20 was available in the dam to sell.) Farmers compete for water that will ultimately determine the quality and quantity of their crop, and in some cases, even the survival of their trees (for example, drought years). It is unlikely that farmers can make credible collusive commitments in such a situation. Contemporaries emphasized the opposite situation: farmers competed aggressively for water, especially during droughts, while water owners were reluctant to lower the price of the water to meet the needs of the poorest farmers.⁶¹

The high number of non-collusive auctions provides evidence farmers did not collude. Farmers met every week, hence the discount rate from one week to the next one was close to 1. If we focus on two consecutive 4-unit auctions, the discount rate is virtually 1. Thus, any collusive agreement would be easy to sustain and we would observe no "price-wars", or deviations from collusive strategies. If the collusion hypothesis were true, all auctions would look collusive except, perhaps, during certain periods where we would observe price-wars. We observe in many cases, however, that both regimes are present during the same week. Unlike Baldwin, Marshall, and Richard (1997) this is not a formal test.⁶²

Nevertheless, taking the analysis one step further, if the collusion hypothesis were true, we would expect more collusion in autumn-winter and less collusion in spring-summer. Incentives

 $^{^{60}\}mathrm{Note}$ also that failure to consider the effect of the structural parameters (SC and DMR) explicitly introduces difficulties.

⁶¹These opinions, along with a qualitative analysis can be found in Vera Nicolás (2004).

⁶²Collusion in repeated auctions has been analyzed conditional (Hopenhayn and Skrzypacz 2004) and unconditional (Porter and Zona 1999 and Pesendorfer 2000) on the history of the game. A discussion on how to detect collusion can be found in Porter (2005).

to deviate from the collusion strategy are higher in spring-summer because the value of the water is higher due to seasonalities (Figures 4 and 7). Punishment is about the same in any season. The maximum punishment would be to play the competitive equilibrium forever. Future discounted earnings in this case are similar in summer and in winter. Hence, deviating from the collusive strategy is more profitable in summer than in winter. However, the data show the opposite pattern. Figure 14 displays the distribution of auctions in the complementarities regime by month. Complementarities are more likely to be observed in summer than in winter, when water requirements (and hence equilibrium prices) soar. This is in line with our interpretation according to the model with sunk and entry costs.

A "competitive" collusion? We have implicitly assumed that farmers' plots were spaced sufficiently far apart from each other. Specifically, we assumed that no other farmer could use the same sub-channel just used by his neighbor. In reality, this assumption is not true for all cases. Because the cost of watering the sub-channel is sunk, if the plots of two farmers are located next to each other and they share the same sub-channel, then one farmer could freeride and outbid the first winner in the second auction. Knowing this, the first winner would bid lower in the first auction. This situation would reduce the revenue of the auction and create inefficiencies. Since farmers might not internalize this free-riding effect, they would take into account the equilibrium outcome for the remaining auctions, and lower their bid in the first auction. They would then will try to outbid their neighbors in later auctions.

In a situation such as this, it would be relatively easy to sustain a collusive agreement among neighboring farmers. The number of members of the coalition would be small (say, three or four farmers), and because they are neighbors, they would know each other well and might even share animals or machinery for agricultural purposes. Each farmer in the coalition would compete in the auction for the first unit, but would not enter the remaining auctions if one member of the coalition won the first unit. With this agreement they would achieve efficiency by solving the free riding problem. With the resulting increase in efficiency, the revenue of the auction would also increase, and the auctioneer would not be opposed to the "collusion". This situation would not affect our results unless farmers coordinated bidding rings to not outbid neighboring farmers in the first auction.⁶³

⁶³It will only affect the outcome when both the bidder with the highest valuation and the bidder with the second highest valuation belong to the same ring, but the bidder with the third highest valuation belongs to a different ring. In this case, our model predicts that the observed price is the valuation of the second highest bidder, but it actually corresponds to the valuation of the third highest bidder. This is unlikely in our empirical setting because the nearly 500 farmers would form around 150 rings (based on the geographical locations that we obtained from the census data). The probability that the two bidders with the highest valuation belong to the same ring is virtually zero. Moreover, the difference between the second highest and the third highest valuation will be small in any case.

⁶⁴There is an extensive literature on the theory of bidding cartels (for example, Graham and Marshall 1987; Hendricks, Porter, and Guofu (2008); Hopenhayn and Skrzypacz (2004); and McAfee and McMillan 1992). For English auctions, Asker (2010) empirically investigates a bidding cartel of collectable stamps. See Harrington (2008) for a survey.

Efficiency. The model displayed in Section 4 assumes that it is costly for the bidders to enter the auction. In order to compare the sequential ascending price auction with other mechanisms, this entry cost has to be taken into account. In this context, and following Stegeman (1996), we interpret entry cost as the cost farmers incur when they send a message to the auctioneer, or to some other farmers. Here, the notions of *ex-ante* and *ex-post* efficiency are no longer equivalent. Although it may be *ex-ante* efficient that more than one player sends a message, it is always *ex-post* efficient that at most one player sends a message.

For this case where it is costly to send messages to the coordinator, Stegeman shows that the ascending price auction has an equilibrium that is *ex-ante* efficient. In contrast, the first-price auction may have no efficient equilibrium, and the author only considers the single-unit case. In our sequential unit case, we have shown that when goods are strict complements the analysis is identical to the single unit case. Hence, the result applies here as well. However, when goods are weak substitutes, the result only applies to the last auction. Although outside the scope of this paper, further work to investigate whether a sequential ascending price auction is *ex-post* efficient when the coordinator has to allocate several objects to players that face SC, DMR, and costly messages, would be a useful extension.

8 Conclusions.

By affecting bidders' behavior in sequential auctions, sunk costs and decreasing marginal returns in the presence of participation costs generate very different price dynamics within the same market. This difference in price dynamics is attributable to the varying extent to which the value of sequential goods complements or falls relative to previous units. The *deterrence* effect, whereby the same bidder pays a high price for the first unit (deterring others from entering subsequent auctions), and a low price for the remaining units, arises when sunk costs are relatively high compared to the decreasing marginal returns, thus creating complementarities among the goods. Substitutability arises due to decreasing returns when sunk costs are relatively small. In this case, equilibrium prices are similar in magnitude, regardless of whether the same or different bidders win the objects. Careful consideration of these features is fundamental to demand characterization, a cornerstone of many positive and normative questions in economics.

Using a novel data set from a decentralized market institution that operated privately for eight centuries in southern Spain, we document these price dynamics and develop a model to recover the underlying structural parameters and distribution of valuations. Although the bidders are better informed than the sellers in our model, the latter know that the sequential English auction allocates water (*ex-ante*) efficiently. Not requiring farmers to reveal their marginal valuations is an advantage of the mechanism, whose simplicity reduces costs associated with its implementation and helps explain its stability. We address three main questions. Are water units complements or substitutes, and why? Is the *deterrence* effect consistent with a competitive market structure or a consequence of collusive behavior among farmers? What would happen to the estimates in this setting if the researcher, by ignoring the importance of participation and sunk costs, failed to account for the complementarity feature of the sequential goods?

First we document that during the period under study both complementarities and substitutabilities are observed in the data, generating different price dynamics. Seasonality, related to the water requirements of the crops and the expected rainfall, affects the relative importance of sunk costs and decreasing returns, causing bidders to coordinate their actions in these regimes. Second, the apparent collusive behavior, when the same bidder wins all the goods, paying very low prices for all the units following the first unit, is actually competitive (or non-cooperative). Contrary to the collusion hypothesis, this behavior is caused by complementarities, and is observed when the value of water (as well as the average price paid per unit and, thus, the incentive to deviate from a collusion strategy) increases relative to the standard competitive pattern registered in the weak substitutes regime. This shows the importance of interpreting the data through the economic model. Finally, by estimating our model, we confirm the relevance of participation and sunk costs in our empirical environment. By testing the performance of our model relative to a standard English auction model without participation costs, we confirm that estimations using the latter are not accurate. Aside from the bias generated by ignoring sunk costs and decreasing returns, price dynamics play an important role, as it is not appropriate to attribute the variation in prices among sequential units (when the goods are complements) to a relatively more disperse underlying distribution of valuations.

References

ABRANTES-METZ, R., S. VILLAS-BOAS, AND G. JUDGE (2011): "Tracking the Libor Rate," Applied Economics Letters.

- ANTON, J. J., AND D. A. YAO (1987): "Second Sourcing and the Experience Curve: Price Competition in Defense Procurement," RAND Journal of Economics.
- ARMSTRONG, M. (2000): "Optimal Multi-Object Auctions," Review of Economic Studies, pp. 455–481.
- ASHENFELTER, O. (1989): "How Auctions Work for Wine and Art," Journal of Economic Perspectives, 3, 23-36.
- ASHENFELTER, O., AND D. GENESOVE (1992): "Testing for Price Anomalies in Real-Estate Auctions," American Economic Review, 82, 501–505.
- ASKER, J. (2010): "A Study of the Internal Organization of a Bidding Cartel," American Economic Review, 100, 724-762.
- ATHEY, S., AND P. A. HAILE (2002): "Identification of Standard Auction Models," Econometrica, 70, 2107–2140.
- ATHEY, S., J. LEVIN, AND E. SEIRA (2011): "Comparing Open and Sealed Bid Auctions: Theory and Evidence from Timber Auctions," Quarterly Journal of Economics, 126, 207–257.
- AUSUBEL, L. M. (2004): "An Efficient Ascending-Bid Auction for Multiple Objects," American Economic Review, 94, 1452–1475.
- BACKUS, M., T. BLAKE, AND S. TADELIS (2013): "Cheap Talk, Round Numbers, and the Economics of Negotiation," Working Paper, Cornell University.

BAKER, M. (1992): "Digit Preference in CPS Unemployment Data," Economics Letters, 39, 117-121.

- BALAT, J. (2013): "Highway Procurement and the Stimulus Package: Identification and Estimation of Dynamic Auctions with Unobserved Heterogeneity," Working Paper, Johns Hopkins University.
- BALDWIN, L. H., R. C. MARSHALL, AND J.-F. RICHARD (1997): "Bidder Collusion at Forest Service Timber Sales," Journal of Political Economy, 105, 657–699.

BENFORD, F. (1938): "The Law of Anomalous Numbers," Proceedings of the American Philosophy Society, 78, 551-572.

BENHARDT, D., AND D. SCOONES (1994): "A Note on Sequential Auctions," American Economic Review, 84.

- BLACK, J., AND D. DE MEZA (1992): "Systematic Price Differences between Successive Auctions Are No Anomaly," Journal of Economics & Management Strategy, 1, 607–28.
- BOPP, M., AND D. FAEH (2008): "End-digits Preference for Self-Reported Height Depends on Language," BMC Public Health, 8, 342.
- BRANCO, F. (1997): "Sequential Auctions with Synergies: An Example," Economic Letters, 54, 159-163.
- CASSADY, R. (1967): Auctions and Auctioneering. University of California Press, Berkeley.
- CHOTT, G., AND L. BRADLEY (1997): "Irrigation Needs of Citrus," Maricopa County (Ariz.) Publication No. MC17.
- COALE, A. J., AND S. LI (1991): "The Effect of Age Misreporting in China on the Calculation of Mortality Rates at Very High Ages," *Demography*, 28, 293–301.
- COMAN, K. (1911): "Some Unsettled Problems of Irrigation," American Economic Review, 1, 1–19.
- CRAMTON, P., AND J. A. SCHWARTZ (2000): "Collusive bidding: Lessons from the FCC spectrum auctions," Journal of regulatory Economics, 17(3), 229–252.
- CRAMTON, P., Y. SHOHAM, AND R. STEINBERG (2006): Combinatorial Auctions. MIT PRess, Cambridge, MA.
- CRAYEN, D., AND J. BATEN (2009): "Numeracy, Inequality, Age Heaping, and Economic Growth: New Estimation Strategies for Western Europe and the U.S. (17th - 19th centuries)," *Economic History Review*.
- DAS GUPTA, P. (1975): "A General Method of Correction for Age Misreporting in Census Populations," *Demography*, 12, 303–12.
- DONALD, S. G., AND H. J. PAARSCH (1996): "Identification, Estimation, and Testing in Parametric Empirical Models of Auctions within the Independent Private Values Paradigm," *Econometric Theory*, 12, 517–567.
- DONNA, J., AND J.-A. ESPIN-SANCHEZ (2012): "Complements and Substitutes in Sequential Auctions: The Case of Water Auctions," Working Paper, The Ohio State University. Available online at http://dl.dropbox.com/u/4794458/JMP_ Donna.pdf.
 - (2013a): "The Illiquidity of Water Markets," Working Paper, The Ohio State University. Available online at https: //dl.dropboxusercontent.com/u/4794458/_WebPage/Papers/LiquidityConstraints.pdf.
- (2013b): "Let the Punishment Fit The Criminal," Working Paper, The Ohio State University. Available online at https://dl.dropboxusercontent.com/u/4794458/_WebPage/Papers/Punishment.pdf.
- DU PREEZ, M. (2001): "Irrigation of Citrus With Reference to Water Shortages and Poor Water Quality," Agricultural Consulters International.
- EDELMAN, B., M. OSTROVSKY, AND M. SCHWARZ (2007): "Internet Advertising and the Generalized Second Price Auction: Selling Billions of Dollars Worth of Keywords," *American Economic Review*, 97, 242–259.

ENGELBRECHT-WIGGANS, R. (1993): "Optimal Auctions Revisited," Games and Economic Behavior, 5, 227–239.

(1994): "Sequential Auctions of Stochastically Equivalent Objects," Economic Letters, 44, 87–90.

- GANDAL, N. (1997): "Sequential Auctions of Interdependent Objects: Israeli Cable Television Licenses," The Journal of Industrial Economics, pp. 227–244.
- GENTZKOW, M. (2007): "Valuing New Goods in a Model with Complementarity: Online Newspapers," American Economic Review, 3, 713–744.
- GIL OLCINA, A. (1994): "Desequilibrio de Recursos Hídricos y Planteamiento de Trasvases en Territorio Valenciano," Coloquio sobre Planificación Hidráulica en España. Instituto Universitario de Geografía. Universidad de Alicante y Fundación Cultural CAM, Alicante.
- GONZÁLEZ-CASTAÑO, J., AND P. LLAMAS-RUIZ (1991): "El Agua en la Ciudad de Mula, S. XVI-XX," Comunidad de Regantes Pantano de la Cierva, Mula, 3.
- GRAHAM, D., AND R. MARSHALL (1987): "Collusive Bidder Behavior at Single-Object Second-Price and English Auctions," Journal of Political Economy, 95, 1217–39.

GROEGER, J. R. (2014): "A Study of Participation in Dynamic Auctions," International Economic Review, 55, 1129-1154.

- HAILE, P. (2001): "Auctions with Resale Markets: An Application to U.S. Forest Service Timber Sales," American Economic Review, 91, 399–427.
- HAILE, P. A., AND E. TAMER (2003): "Inference with an Incomplete Model of English Auctions," *Journal of Political Economy*, 111, 1–51.

- HARRINGTON, JOSEPH E., J. (2008): "Detecting Cartels," In: Handbook of Antitrust Economics, ed. Paolo Buccirossi, Cambridge, MA: MIT Press, pp. 213–58.
- HENDRICKS, K., AND R. H. PORTER (1988): "An Empirical Study of an Auction with Asymmetric Information," American Economic Review, 78, 865–883.
- HENDRICKS, K., R. H. PORTER, AND T. GUOFU (2008): "Bidding Rings and the Winner's Curse: The Case of Federal Offshore Oil and Gas Lease Auctions," *RAND Journal of Economics*, 4, 1018–1041.
- HICKMAN, B. R., T. P. HUBBARD, AND Y. SAGLAM (2011): "Structural Econometric Methods in Auctions: A Guide to the Literature," Journal of Econometric Methods, 1.
- HOPENHAYN, H., AND A. SKRZYPACZ (2004): "Tacit Collusion in Repeated Auctions," Journal of Economic Theory, 114, 153–169.
- HORTAÇSU, A. (2011): "Recent Progress in the Empirical Analysis of Multi-Unit Auctions," International Journal of Industrial Organization, 29, 345–349.
- JEITSCHKO, T., AND E. WOLFSTETTER (2002): "Scale Economies and the Dynamics of Recurring Auctions," *Economic Inquiry*, 40, 403–414.

JOFRE-BONET, M., AND M. PESENDORFER (2003): "Estimation of a Dynamic Auction Game," Econometrica, pp. 1443–1489.

— (2012): "Optimal Sequential Auctions," Mimeo, LSE.

- KAGEL, J. H. (1995): "Auctions: A Survey of Experimental Research," in J. Kagel and A. Roth, eds., Handbook of Experimental Economics. Princeton, N.J.: Princeton University Press, pp. 501–85.
- KAGEL, J. H., AND D. LEVIN (2001): "Behavior in Multi-Unit Demand Auctions: Experiments with Uniform Price and Dynamic Vickrey Auctions," *Econometrica*, 69, 413–454.
- (2005): "Multi-Unit Demand Auctions with Synergies: Behavior in Sealed-Bid versus Ascending-Bid Uniform-Price Auctions," *Games and Economic Behavior*, 53, 170–207.
- KANDEL, S., O. SARIG, AND A. WOHL (2001): "Do Investors Prefer Round Stock Prices? Evidence From Israeli IPO Auctions," Journal of Banking & Finance, 25, 1543–1551.
- KASTL, J. (2011): "Discrete Bids and Empirical Inference in Divisible Good Auctions," Review of Economic Studies, 78, 974–1014.
- KRASNOKUTSKAYA, E. (2011): "Identification and Estimation in Highway Procurement Auctions under Unobserved Auction Heterogeneity," Review of Economic Studies, 78, 293–327.

LAFFONT, J.-J., H. OSSARD, AND Q. VUONG (1995): "Econometrics of First-Price Auctions," Econometrica, 63, 953-980.

- LARSEN, B. (2013): "The Efficiency of Dynamic, Post-Auction Bargaining: Evidence from Wholesale Used Auto Auctions," Working Paper, Stanford University.
- LEVIN, J. (1997): "An Optimal Auction for Complements," Games and Economic Behavior, 18, 176–192.
- LIST, J., D. MILLIMET, AND M. PRICE (2004): "Inferring Treatment Status when Treatment Assignment is Unknown: with an Application to Collusive Bidding Behavior in Canadian Softwood Timber Auctions," *Technical report, University of Maryland.*
- LIU, Q. (2011): "Equilibrium of a Sequence of Auctions when Bidders Demand Multiple Items," Economic Letters, 112, 192–194.
- MARSHALL, R., M. RAIFF, J.-F. RICHARD, AND S. SCHULENBERG (2006): "The Impact of Delivery Synergies on Bidding in the Georgia School Milk Market," The B.E. Journal of Economic Analysis & Policy, 6, 1–51.
- MCAFEE, R. P., AND J. MCMILLAN (1987): "Auctions with Entry," Economic Letters, 23, 343-7.
- (1992): "Bidding Rings," American Economic Review, 82, 579–99.
- MCAFEE, R. P., AND D. VINCENT (1993): "The Declining Price Anomaly," Journal of Economic Theory, 60, 191-212.
- MILGROM, P. (2000): "Putting Auction Theory to Work: The Simultaneous Ascending Auction," The Journal of Political Economy, pp. 245–272.
- MILGROM, P., AND R. WEBER (1982): "A Theory of Auctions and Competitive Bidding," Econometrica, 50, 1089-1122.
- MYERS, R. (1940): "Errors and Bias in the Reporting of Ages in Census Data," Transactions of the Actuarial Society of America, 41, 395–415.
- OSTROM, E. (1992): Crafting Institutions for Self-Governing Irrigation Systems. ICS Press, San Francisco, CA.
- PALFREY, T. (1983): "Bundling Decisions by a Multiproduct Monopolist with Incomplete Information," *Econometrica*, pp. 463–484.

PESENDORFER, M. (2000): "A Study of Collusion in First Price Auctions," Review of Economic Studies, 67, 381-411.

PORTER, R. H. (2005): "Detecting Collusion," Review of Industrial Organization, 26, 147-167.

- PORTER, R. H., AND J. D. ZONA (1999): "Ohio School Milk Markets: An Analysis of Bidding," RAND Journal of Economics, 30, 263–288.
- RAUCH, B., M. GOETTSCHE, G. BRAEHLER, AND S. ENGEL (2011): "Fact and Fiction in EU-Governmental Economic Data," German Economic Review, 12, 243–255.
- REGUANT, M. (2013): "Complementary Bidding Mechanisms and Startup Costs in Electricity Markets," Working Paper, Stanford University.

ROBERTS, J. (2009): "Unobserved Heterogeneity and Reserve Prices in Auctions," Working Paper, Duke University.

- SIEGEL, J. S., H. SHRYOCK, AND D. SWANSON (2003): The Methods and Materials of Demography. Academic Press Inc, Amsterdam, second edn.
- Song, U. (2004): "Nonparametric Estimation of an eBay Auction Model with an Unknown Number of Bidders," Mimeo, University of British Columbia.
- STEGEMAN, M. (1996): "Participation Costs and Efficient Auctions," Journal of Economic Theory, 71, 228–259.
- VALIENTE, O. (2001): "Sequía: Definiciones, Tipologías y Métodos de Cuantificación," Investigaciones Geográficas, 26, 59-80.
- VERA NICOLÁS, P. (2004): Murcia y el Agua: Historia de Una Pasión. Academia Alfonso X El Sabio. Available online at http://servicios.laverdad.es/murcia_agua/, Murcia.

VON DER FEHR, N.-H. (1994): "Predatory Bidding in Sequential Auctions," Oxford Economic Papers, 46, 345–356.

- WEBER, R. (1983): "Multiple-Object Auctions," in R. Engelbrecht-Wiggans, R.M. Stark, M. Shubik., eds., Auctions, Bidding and Contracting, New York: NYU Press, pp. 165–691.
- WOLFRAM, C. (1998): "Strategic Bidding in a Multi-Unit Auction: an Empirical Analysis of Bids to Supply Electricity in England and Wales," *RAND Journal of Economics*, pp. 703–725.
- WRIGHT, G. (2000): "Irrigating Citrus Trees," The University of Arizona, College of Agriculture, Publication No. AZ1151, 46, 1–5.
- ZULEHNER, C. (2009): "Bidding Behavior in sequential Cattle Auctions," International Journal of Industrial Organization, 27, 33–42.

Auction $\#$	Name	Price	Day	
1	Pedro Fernández	123		
2	Pedro Fernández	111	Мо	
3	Pedro Fernández	111	MO	DIA Primer ()
4	Pedro Fernández	109		1 pilata Jama 182
5	Pedro Blaya	115		2 · f. Annone /// - . A filment
6	Jose Ruiz	116		4 - 2 parting 163
7	Mauricio Gutiérrez	117	Iu	6 - Tore Pier
8	Mauricio Gutiérrez	106		n d lignerer 111-
9	Ambrosio Ortíz	116		9 - Machine Polar 116 10 - Il parme 100 -
10	Ambrosio Ortíz	100	Wo	11 - Il anna In In
11	Ambrosio Ortíz	100	we	13 - Elinen Janeur It.
12	Carlota Pomares	116		13 - Acture sprane 110 -
13	Eliseo Gutiérrez	120		17 - The plane 111
14	Antonio Muñoz	112	Th	19 - Jaan Mathim 91-
15	Antonio Navarro	110	111	20 - Torre fatimer 100-
16	Vicente Ledesma	106		Die 2772- Nache 1564
17	Jose Gálvez	103		Somer JFII
18	Juan Martínez	91	Fr	To al republic
19	Juan Martínez	90	L LI	monorma mar 18 m februs de 1805
20	Jesus Gutiérrez	100		1 decemp

Figure 1: Auction Sample: Goods Are Substitutes

Sample from original data obtained from the historical archive: Goods Are Substitutes (Winter - February 18, 1955, Day). Units 1 to 4 are the units bought on Monday (Mo) during day (unit 1 corresponds to right to irrigate from 7AM to 10AM, unit 2 from 10AM to 1PM, unit 3 from 1PM to 4PM, and unit 4 from 4PM to 7PM). Similarly, units 5 to 8 are the units bought on Tuesday (Tu) during day; units 9 to 12 are the units on Wednesday (We) during day; units 13 to 16 are the units on Thursday (Th) during day; and units 17 to 20 are the units on Friday (Fr) during day.

Figure 2: Auction Sample: Goods Are Complements

Auction $\#$	Name	Price	Day	DIA	Printer Co.
1	Juana Fernández	1580		A. Jerus der	1580 -
2	Juana Fernández	50	١	2 la minune	50 -
3	Juana Fernández	50	MO	3 for human	50-
4	Juana Fernández	50		3 pravices fabarran	1401 -
5	Francisco Gabarrón	1401		7 _ d pinno	50-
6	Francisco Gabarrón	50	.	9. Jore from	1401 -
7	Francisco Gabarrón	50	l	10 il Juino	25-
8	Francisco Gabarrón	50		12 - A accuss Mature	25-
9	Jose Fernández	1401		14 . Il durino	25-
10	Jose Fernández	25	337-	15 d min	25 -
11	Jose Fernández	25	we	17 Manuel fature	1406 -
12	Jose Fernández	25		19 - il suismo	50 -
13	Antonio Belijar Boluda	1401		20 · · · · · · · · · · · · · · · · · · ·	7789 -
14	Antonio Belijar Boluda	25	D 1	Noche	5929-
15	Antonio Belijar Boluda	25	ln	Voz pública	1-
16	Antonio Belijar Boluda	25		Total líquido	3717-
17	Manuel Gutiérrez	1406		Mula 22 de Julio de 19. Ro. U Secretorio.	66
18	Manuel Gutiérrez	50	D .		
19	Manuel Gutiérrez	50	Fr		
20	Manuel Gutiérrez	50			
	1		1		

Sample from original data obtained from the historical archive: Goods Are Complements (Summer - July 22, 1966, Day). See notes in Figure 1.

Figure 3: Sample of Individual Data Obtained from the Agricultural Census



Sample Card from a Farmer Obtained from the Agricultural Census. Individual characteristics include: farmers' name (that we match to the names in the auctions), type of land and location, area, number of trees, production and the price at which this production was sold in the census year.

Figure 4: Rain and Frequency Distribution of 4CU Over the Sample Period



The figure displays for each month: i) the number of auctions where the same farmer wins all four consecutive units, and ii) total rain using a Nadaraya-Watson kernel regression (of 'total rain' on 'month of the year') with an Epanechnikov kernel with bandwidth selected by cross validation.



Figure 5: Winning Prices: by Number of Consecutive Units Bought by the Same Farmer and by Unit



The figure displays the distribution of winning prices by: i) Unit (First Unit in Blue, Second Unit in Red, Third Unit in Green, and Fourth Unit in Orange); Weekday (Mo=Monday, Tu=Tuesday, We=Wednesday, Th=Thursday, and Fr=Friday); and Schedule (Day=Day-Time and Night=Night-Time). Thus, the figure displays the distribution of prices of each of the 40 units auctioned per week for the whole sample (disaggregated by Unit, Weekday, and Schedule). Each vertical box (unit) displays the maximum price (upper adjacent value), 75th percentile (upper hinge), median (black circle marker), 25th percentile (lower hinge), and minimum price (lower adjacent value).





The figure displays the distribution of winning prices by season. Each vertical box displays the maximum price (upper adjacent value), 75th percentile (upper hinge), median (black circle marker), 25th percentile (lower hinge), and minimum price (lower adjacent value).





The figure displays the distribution of winning prices by: i) Season and Drought Indicator. Each vertical box displays the maximum price (upper adjacent value), 75th percentile (upper hinge), median, 25th percentile (lower hinge), and minimum price (lower adjacent value). We define a drought as an indicator that equals one when average monthly rain during the specific year is below a consensus threshold defined in the literature in terms of the historic annual average (following Gil Olcina 1994 we use a threshold of 40%). The numbers below each box correspond to the percentage (in terms of the whole sample) of observations in each box (*i.e.* al these numbers sum up to 100%).

Figure 9: The Channel System in Mula and the Sunk Cost of Initiating the Irrigation





The main canal (left panel) was made of concrete. The individual sub-channels (right panel) were dug into the ground. Thus, in these sub-channels, a water loss is incurred because water flows over a dry sub-channel (some water is absorbed by the ground).



Figure 10: Marginal Returns of Irrigation Water

Marginal returns of water in summer (left) and autumn (right).

Figure 11: Within Price Distribution by Regime: First Price vs. Median Second to Fourth prices



Both figures display, for instances when the same farmer won all four units in a 4-unit auction, the normalized percentage change, Δ , of the first winning price against the median of the second to fourth winning prices: $\Delta = \frac{p_1 - m}{(p_1 + m)/2}$, with $m = median (p_2, p_3, p_4)$. Note that $\Delta < 2$ and that $\Delta \rightarrow 2$ if, and only if, $p_1 \rightarrow \infty$, or $m \rightarrow 0$, or both; *i.e.* when the percentage change goes to infinity. In the figure on the left, Lower Bound is computed by assuming that all second, third, and fourth unit prices paid by the (same) farmer within the same four-unit auction display end-price preference for 0. In the figure on the right, Upper Bound is computed by assuming that only one among second, third, or fourth unit prices paid by the (same) farmer within the same four-unit auction display end-price preference for 0.

Auction $\#$	Name	Price	Day	
1	Sebastian Aguilar	48		DIA
2	Felipe Amaro	42	Ъſ	Dilebartion aquilar
3	Felipe Amaro	48	MO	3 d Minuo 42.
4	Diego Guirao	50		5 felipe June 50 -
5	Felipe Amaro	54		7 . Curtolal Rouers 47.
6	Antonio Llamas	51	-	9 - Distubel future 2.
7	Cristóbal Romero	47	Iu	10 · d allerens 5.
8	Cristóbal Romero	50		12 " el dur 1 - 13 · funo allega 2 75
9	Cristóbal Gutiérrez	2		14 ° el manno 1 -
10	Cristóbal Gutiérrez	5		16 · el recircuo 1 ·
11	Cristóbal Gutiérrez	1	We	
12	Cristóbal Gutiérrez	1		33 .) Dia
13	Luis Moya	2.75		Noche
14	Luis Moya	1	T 1	Voz pública <u>I e e</u> To al liquido <u>523</u>
15	Luis Mova	1	Th	11 22 de Juno de 195 y
16	Luis Moya	1		El Presidente, A dester
17				1. Jaim
18			D	Survey Annual
19			Fr	
20				12 Carl 10 Carl

Figure 12: Auction Sample: Auction where Farmers Are Present and No Bids Are Placed

Sample from original data obtained from the historical archive: Auction where farmers are present and no bids are placed (Winter - January 22, 1954, Day).



Figure 13: Winning and Estimated Prices

The figure displays real prices against predicted prices using three different models: (i) our structural model (specification 3 in Table 7), (ii) a standard (button) English auction model (specification 5 in Table 7), and (iii) a reduced-form model for the sample using as regressors: *Past Rain*, unit (3 dummy variables), weekday (4 dummy variables), schedule (1 dummy variable), month (11 dummy variables), year (12 dummy variables), and individual fixed effects, in addition to a constant (for details about the reduced-form specification see Table 2 discussed in Subsection 3.3). The graph shows the mean monthly averages of the prices. Similar results are obtained using a spline (available in our earlier working paper Donna and Espin-Sanchez 2012). See Section C in the online appendix for a high definition version of this figure.



Figure 14: Regime Frequency Disaggregation by Month

The figure depicts the frequency of auctions where the same farmer buys all four consecutive units (4CU), by regime (see Section 3) and month. (Note that the sum of 4CU over months and regimes—the vertical lines in the graph—is equal to 1470 = 5880/4. See Table 1 in the paper and Table A1 in the online appendix.) It can be seen that complementarities are more likely to be observed in summer than in winter, where water requirements (and, hence, equilibrium prices) soar. We interpret this as evidence in favor of the competition hypothesis (according to our model with entry and sunk costs) and against the collusion hypothesis.

Table 1: Distribution of Winning Prices: by Number of CU and Sequential Auction

ranei 1: rice distribution by number of consecutive bids. All Auctions										
	Median	Mean	SD	Max	Min	Obs				
1CU	101	218.2	327.9	3000	0.05	3530				
$2\mathrm{CU}$	123	256.7	364.6	2700	0.05	2866				
$3\mathrm{CU}$	190	320.0	415.5	4050	0.05	1716				
$4\mathrm{CU}$	182	339.9	470.2	4830	0.05	5880				

Panel 1: Price distribution by number of consecutive bids: All Auctions

Panel 2: Price distribution by number of consecutive bids: First Auction

	Median	Mean	SD	Max	Min	Obs
1CU	100	211.1	304.1	2921	0.05	977
$2\mathrm{CU}$	150	305.0	427.8	2700	0.05	673
$3\mathrm{CU}$	220.5	410.0	512.5	4050	0.05	382
$4\mathrm{CU}$	451	677.6	689.5	4830	0.05	1470

Panel 3: Price distribution by number of consecutive bids: Second Auction

	Median	Mean	SD	Max	Min	Obs
1CU	93.25	219.8	373.0	3000	0.10	624
$2\mathrm{CU}$	103.5	230.2	328.0	2685	0.05	867
$3\mathrm{CU}$	181	294.9	364.7	2850	0.05	539
$4\mathrm{CU}$	101	242.7	309.3	2605	0.05	1470

Panel 4: Price distribution by number of consecutive bids: Third Auction

	Median	Mean	SD	Max	Min	Obs
1CU	94	200.8	312.3	2357	0.10	715
$2\mathrm{CU}$	126.5	256.5	353.9	2601	0.10	778
$3\mathrm{CU}$	151.5	285.55	379.0	2801	0.05	536
$4\mathrm{CU}$	100	229.2	294.2	2701	0.05	1470

Panel 5: Price distribution by number of consecutive bids: Fourth Auction

	Median	Mean	SD	Max	Min	Obs
1CU	114.5	233.4	330.2	2601	0.05	1214
$2\mathrm{CU}$	113.5	239.6	344.8	2601	0.10	548
$3\mathrm{CU}$	167	311.6	411.6	2630	0.05	259
$4\mathrm{CU}$	100	210.1	272.6	2935	0.05	1470

Panel 6: Price distribution for 4CU

Auction	Median	Mean	SD	Max	Min	Obs
1st to 4th	182	339.9	470.2	4830	0.05	5880
1st	451	677.6	689.5	4830	0.05	1470
2nd	101	242.7	309.3	2605	0.05	1470
3rd	100	229.2	294.2	2701	0.05	1470
4th	100	210.1	272.6	2935	0.05	1470
1st and 2nd	253	460.2	576.9	4830	0.05	2940
2nd and 3rd	101.0	235.9	301.9	3001	0.05	2940
3rd and 4th	100.0	219.7	283.7	2935	0.05	2940
1st to 3rd	200.0	383.2	512.4	4830	0.05	4410
2nd to 4th	100.0	227.3	292.7	3001	0.05	4410

The table displays the Distribution of Winning Prices. Panels 1 to 5 presents the Distribution of Prices disaggregated by cases where the same farmer buys one, two, three, or four consecutive units (1CU, 2CU, 3CU, or 4CU, respectively). Panel 1 presents the Distribution of Prices for All Auctions (*i.e.* First, Second, Third, and Fourth Auctions). Panel 2 presents the Distribution of Prices for First Auctions. Panel 3 presents the Distribution of Prices for Second Auctions. Panel 4 presents the Distribution of Prices for Third Auctions. Panel 5 presents the Distribution of Prices for Fourth Auctions. Finally, Panel 6 presents Distribution of Prices just for 4CU (*i.e.* for the subsample of 5880 auctions where the same farmer won all four consecutive units). Note that the first line in Panel 6 (1st to 4th) displays the same information as the last line in Panel 1 (4CU). The second line in Panel 6 (1st) displays the same information as the last line in Panel 6 (2nd) displays the same information as the last line in Panel 6 (2nd) displays the same information as the last line in Panel 6 (4CU). The fifth line in Panel 3 (4CU). The fourth line in Panel 6 (3rd) displays the same information as the last line in Panel 5 (4CU).

Variables	(1)	(2)	(3)	(4)
Rain MA7	-4.0543***	-4.1117***	-2.9911***	-3.1741***
	(0.6742)	(0.6894)	(0.5580)	(0.6227)
Rain Day Bought	-0.2346	-0.1853	0.0519	0.1779
	(0.1434)	(0.1416)	(0.1558)	(0.1531)
Unit 2 Day		-167.9547***	-167.8286***	-180.6172***
		(19.4659)	(19.4542)	(21.8896)
Unit 3 Day		-173.0328***	-172.9066***	-188.0507***
		(19.9287)	(19.9165)	(22.7544)
Unit 4 Day		-176.5446***	-176.5451***	-190.8276***
		(20.4404)	(20.4387)	(23.0968)
Unit 2 Night		-237.5795***	-237.8597***	-249.3275***
-		(24.9031)	(24.9367)	(27.3493)
Unit 3 Night		-243.3244***	-243.5220***	-257.6533***
-		(25.4507)	(25.4838)	(28.3077)
Unit 4 Night		-254.8376***	-255.1867***	-266.4109***
		(25.7254)	(25.7817)	(28.9070)
Tuesday		26.0232***	32.1906***	10.0596
		(7.4927)	(8.2359)	(12.4758)
Wednesday		-34.5756***	-29.3838**	-31.7270**
		(10.6269)	(11.9714)	(15.4702)
Thursday		-59.6530***	-55.4057***	-42.9164***
		(10.7704)	(12.3518)	(15.0439)
Friday		-94.9538***	-95.5421***	-76.3654^{***}
		(12.7055)	(14.4995)	(17.2939)
Night		-110.0908***	-111.0406***	-102.6780***
		(11.3642)	(10.9544)	(13.4322)
Unit FE	No	Yes	Yes	Yes
Weekday FE	No	Yes	Yes	Yes
Schedule FE	No	Yes	Yes	Yes
Month FE	No	No	Yes	Yes
Individual FE	No	No	No	Yes
R^2	0.016	0.083	0.230	0.359
Observations	$13,\!801$	$13,\!801$	13,801	$13,\!801$

Table 2: Correlation Between Winning Prices and Covariates

All columns are OLS regressions. Dependent variable is the winning price in each auction (one *cuarta*). Robust standard errors in parentheses. FE stands for *Fixed Effects. Individual FE* refers to a set of dummy variables identifying different winners (names) in our sample. We obtain similar results including *Week FE* (a set of dummy variables identifying 52 or 53 weeks of the corresponding year). *** p<0.01, ** p<0.05, * p<0.1. Sample restricted to auctions with positive bids during the period January 1954 to August 1966.

	All u	nits	First	Unit	Non First Unit		Second Unit		Third Unit		Fourth Unit		
	(1)	(2)	(3	(3)		(4)		(5)		(6)	
Last Digit	Freq	%	Freq	%	Freq	%	Freq	%	Freq	%	Freq	%	
0	2,905	49.9	468	32.1	$2,\!437$	55.8	809	55.5	811	55.7	817	56.1	
1	1,541	26.4	469	32.2	1,072	24.5	370	25.4	369	25.3	333	22.9	
2	201	3.5	116	8.0	85	1.9	32	2.2	24	1.7	29	2.0	
3	138	2.4	59	4.0	79	1.8	29	2.0	23	1.6	27	1.9	
4	68	1.2	40	2.7	28	0.6	10	0.7	9	0.6	9	0.6	
5	531	9.1	123	8.4	408	9.3	114	7.8	144	9.9	150	10.3	
6	218	3.7	91	6.2	127	2.9	43	3.0	37	2.5	47	3.2	
7	71	1.2	33	2.3	38	0.9	19	1.3	12	0.8	7	0.5	
8	61	1.0	24	1.6	37	0.9	8	0.6	12	0.8	17	1.2	
9	48	0.8	24	1.6	24	0.6	12	0.8	5	0.3	7	0.5	
Total	5,828	100	$1,\!457$	100	4,371	100	$1,\!457$	100	1,457	100	$1,\!457$	100	

Table 3: End-Price Preferences

Sample restricted to 4CU auctions (*i.e.* instances when the same farmer won all four units in a 4-unit auction). Last Digit refers to the end-digit winning price. Non integer winning prices are excluded.

		Fir	st Unit	-	Fourth Unit				
	Strict	Compl	Weak	Substitutes	Strict Compl		Weak Substitutes		
	(1	1)		(2)	(3	3)		(4)	
Last Digit	Freq	%	Freq	%	Freq	%	Freq	%	
0	220	38.3	236	29.0	575	100	240	29.5	
1	185	32.2	264	32.5	0	0	323	39.7	
2	42	7.3	69	8.5	0	0	24	2.9	
3	17	3.0	41	5.0	0	0	26	3.2	
4	16	2.8	22	2.7	0	0	6	0.7	
5	45	7.8	76	9.4	0	0	125	15.4	
6	34	5.9	50	6.2	0	0	43	5.3	
7	8	1.4	22	2.7	0	0	6	0.7	
8	6	1.0	14	1.7	0	0	13	1.6	
9	2	0.4	19	2.3	0	0	7	0.9	
Total	575	100	813	100	575	100	813	100	

Table 4: End-Price Preferences by Regime

Sample restricted to 4CU auctions. *Last Digit* refers to the end-digit winning price. Non integer winning prices are excluded. Regime is determined by assuming all second, third, and four unit prices paid by the same winner within the same four-unit auction display end-price preference for 0.

Month	1954	1955	1956	1957	1958	1959	1960	1961	1962	1963	1964	1965	1966	Total
1	4	6	0	3	1	11	3	0	0	0	28	0	11	61
2	4	4	0	2	4	2	3	0	0	0	19	0	21	57
3	5	3	0	9	0	1	2	0	0	10	29	8	23	79
4	0	2	0	6	2	5	4	6	0	17	28	38	28	121
5	5	7	0	6	1	13	9	6	9	4	32	30	31	130
6	3	7	0	7	0	8	10	7	10	14	23	25	29	119
7	2	3	0	6	9	26	8	5	13	15	17	21	23	117
8	9	3	0	3	4	10	7	14	18	15	21	16	3	102
9	8	8	0	3	8	10	5	13	0	8	35	19	0	97
10	8	7	3	2	11	2	0	9	0	10	16	19	0	78
11	7	2	3	0	8	2	0	4	0	21	29	23	0	82
12	1	0	2	2	3	1	0	0	0	36	18	12	0	69
Total	48	43	8	43	48	80	47	54	44	106	179	147	128	537

Table 5: Timing Structure of Different Winners: Estimation Sample

Total, in the last row, refers to the total number of *different winners* for the *specific year* (column). Given that, within a year, the same bidders win multiple units in several months, this number is below the sum over months, by year. Similarly for the last column, where *Total* is the number of *different bidders* for the *specific month* (row) during the 13-year sample. Finally, 537, refers to the total number of different bidders in the whole sample. The monthly average of different bidders who bought water in the sample (years 1954 to 1966) is 8.31.

Table 6: Rain Expectations and Regime Coordination

Variables	(1)	(2)	(3)
Future Rain	-0.0030*** (0.0013)	-0.0030*** (0.0013)	$\begin{array}{c} -0.0034^{***} \\ (0.0013) \end{array}$
Weekday FE	NO	YES	YES
Schedule FE	NO	YES	YES
Month FE	NO	NO	YES

Sample restricted to the one used in the structural estimation in Table 7. Almost identical results are obtained using the whole sample. All specifications are probit regressions. Marginal effects are reported. Robust standard errors in parenthesis. Dependent variable is a dummy variable equal to one if the regime is strict complements (see Section 5). *Future Rain* is a moving average of rain in Mula for seven days after the corresponding date of the auction (*Future Rain* is a proxy variable for farmers' rain expectations for the day where they are buying water). *Past Rain* (a moving average of rain in Mula for seven days before the corresponding date of the auction) and *Actual Rain* (the amount of rain in Mula in the day of the auction) are not statistically significant in any of the above regressions. *** p < 0.01, ** p < 0.05, * p < 0.1.

			Specif	ications		
Structural parameters		Sequential A	Auction Mo	del	Standar	d Model
	(1)	(2)	(3)	(4)	(5)	(6)
Moon Voluction [E(V)]	148.25	127.11	152.93	138.94	166.99	142.73
Mean valuation $[\mathbb{E}(V)]$	(10.129)	(8.153)	(70.832)	(61.847)	(87.373)	(81.648)
Sunk Cost $(\hat{\alpha})$	0.0301	0.0300	0.0301	0.0303		
Sunk Cost (a)	(8.2e-04)	(9.7e-04)	(3.0e-03)	(9.1e-03)		
Âc	0.0101	0.0103	0.0102	0.0105		
ρ_0	(1.2e-05)	(1.3e-04)	(4.9e-03)	(1.3e-03)		
\hat{eta}_0^s	0.0100	0.0110	0.0101	0.0112		
	(0.0024)	(0.0020)	(4.9e-03)	(1.3e-03)		
Future Rain						
$\hat{\beta}_1^c$	5.99e-10	1.54e-10	5.29e-10	5.94 e- 10		
-	(1.6e-04)	(1.5e-04)	(1.1e-10)	(3.2e-10)		
$\hat{\beta}_1^s$	0.21279	0.18280	0.21278	0.18282		
· 1	(0.0059)	(0.0037)	(0.0267)	(0.0775)		
				. ,		
Mean $\hat{\rho}$						
- Strict Complements	0.3801	0.2456	0.6811	0.4239		
- Weak Substitutes	-0.3056	-0.2655	-0.3085	-0.2637		
N	8	10	8	10	8	10
Past Rain Polynomial	No	No	Yes	Yes	Yes	Yes
Schedule Dummy	No	No	Yes	Yes	Yes	Yes
Weekday Dummy Variables	No	No	Yes	Yes	Yes	Yes
Month Dummy Variables	No	No	Yes	Yes	Yes	Yes
Pseudo R^2	0.2832	0.2751	0.5303	0.4842	0.1414	0.1333
Log likelihood	-12,870	-13,940	-11,423	-12,375	-50,930	-54,956
# of Auctions	$5,\!951$	$5,\!951$	$5,\!951$	$5,\!951$	$5,\!951$	$5,\!951$

 Table 7: Structural Estimation

Bootstrapped standard errors (B = 1,000) are reported in parenthesis (for the Mean Valuation it corresponds to the bootstrapped standard error corresponding to $Z'_t \gamma$). Estimates in columns 1 to 4 (sequential auction model) are obtained using the estimation procedure described in Subsection 6.1 using KNITRO, a solver for nonlinear optimization, with tolerance level of 1.0e-12 (see Sections D and E in the online appendix for details). For the distribution of private values and inclusion of covariates, we use an exponential distribution. Estimates in specifications 5 and 6 (standard model) are MLE obtained by maximizing the likelihood function from a standard English auction model allowing the mean of the distribution of valuations depend on the same characteristics as in the other specifications as indicated in equation 7, without fixed costs nor decreasing marginal returns (the sample is the same as the one in columns 1 to 4, including in this case all sequential prices in the estimation). Number of years in the sample is 13. Number of months in the sample is 119. The number of different winners (across all 13 years) is 537. The complementarity parameter, ρ , is computed as detailed in the in Section 4. When the goods are strict complements is given by $\rho_t^C = \frac{\alpha - 3(\hat{\beta}_0^C + \hat{\beta}_1^C T_t)}{1-\alpha} = \frac{\alpha - 3(\hat{\beta}_0^C + \hat{\beta}_1^C T_t) D_{a,t}R_t^F)}{1-\alpha}$. Similarly, when the goods are weak substitutes, the table reports, for each specification: $\hat{\rho}^S = \frac{\hat{\alpha} - 3(\hat{\beta}_0^S + \hat{\beta}_1^C T_t) (\sum_{t=1}^T (D_{b,t}R_t^F + D_{c,t}R_t^F))}{1-\hat{\alpha}}$.

Structural parameters	Specifications Sequential Auction Model Standard Model						
1	(1)	(2)	(3)	(4)	(5)	(6)	
Covariates							
Past Rain $(\hat{\gamma}_1)$			-1.668	-2.076	-1.416	-1.2422	
(/1)			(0.301)	(0.547)	(0.379)	(0.664)	
$(\text{Past Rain})^2 (\hat{\gamma}_2)$			0.0076	0.0126	0.0034	0.0032	
			(0.0448)	(0.0958)	(0.1496)	(0.1520	
Night $(\hat{\gamma}_3)$			-27.250	-23.30	-30.003	-20.910	
			(1.000)	(9.049)	(5.039)	-2 /229	
Tuesday $(\hat{\gamma}_4)$			(0.6449)	(9,7063)	$(7\ 1630)$	(2.6197)	
(.)			-2.4616	-2.1883	-5.8751	-5.0944	
Wednesday $(\hat{\gamma}_5)$			(0.5406)	(0.9572)	(2.5602)	(6.0395)	
			-8.8423	-8.366	-15.755	-13.502	
1 nursday (γ_6)			(0.6169)	(0.8544)	(4.7014)	(2.5863)	
Friday (â.)			-17.805	-9.795	-28.154	-24.271	
riday (77)			(5.016)	(3.418)	(12.903)	(2.118)	
Feb $(\hat{\gamma}_{e})$			-11.373	-41.023	-4.8293	-5.0584	
100. (78)			(23.299)	(35.705)	(2.330)	(4.949)	
Mar. $(\hat{\gamma}_9)$			27.356	18.067	34.954	30.126	
(10)			(12.386)	(8.491)	(5.218)	(1.676)	
Apr. $(\hat{\gamma}_8)$			82.481	53.902	(8.799)	07.390	
,			(23.430) 115 140	(17.127) 81.751	(3.420) 114 483	(11.579	
May. $(\hat{\gamma}_{10})$			(24.822)	(30.187)	(4.366)	90.142 (15.015	
			(24.022) 49.771	40 248	(4.300) 57 478	48 341	
Jun. $(\hat{\gamma}_{11})$			(23.112)	(18.698)	(8.429)	(15.16)	
T 1 (^)			195.980	115.752	225.337	191.341	
Jul. (γ_{12})			(26.035)	(48.982)	(216.025)	(102.962)	
A_{110} $(\hat{\alpha}_{11})$			233.750	183.06	247.57	210.08	
Aug. (713)			(28.562)	(27.608)	(195.347)	(87.953)	
Sep $(\hat{\gamma}_{14})$			74.494	160.78	88.173	74.259	
~~p. (/14)			(23.684)	(34.040)	(39.318)	(12.133	
Oct. $(\hat{\gamma}_{15})$			77.623	62.664	81.093	69.953	
() = =) /			(14.532)	(22.385)	(30.165) 12.2000	(10.558	
Nov. $(\hat{\gamma}_{16})$			3.3022 (2.3041)	(7513)	15.2099 (16.725)	10.0131 (0.6074	
			(2.3941) 7 4696	-0.9495	(10.725) 2 8462	2 2106	
Dec. $(\hat{\gamma}_{17})$			(2, 2226)	(25513)	(3.6735)	(4.0308)	
	148.253	127.117	90.885	96.131	(0.0100) 101.717	87.978	
Intercept $(\hat{\gamma}_0)$	(10.121)	(8.153)	(32.274)	(25.754)	(179.285)	(74.073	
N	8	10	8	10	8	10	
Pseudo R^2	0.2832	0.2751	0.5303	0.4842	0.1414	0.1333	
Log likelihood	-12,870	-13,940	-11,423	-12,375	-50,930	-54,956	
# of Auctions	$5,\!951$	5,951	5,951	$5,\!951$	$5,\!951$	$5,\!951$	

 Table 8: Structural Estimation (continued)

See notes in Table 7.